Massey University, Auckland, New Zealand, and Gao liang Tian School of Management, Xi'an Jiaotong University, Xi'an, China

Aggressive tax planning and stock

price synchronicity: evidence

from China

Hua Feng Department of Accounting and Finance, Xi'an Jiaotong University, Xi'an, China

Ahsan Habib

Abstract

Purpose – The purpose of this paper is to investigate the association between aggressive tax planning and stock price synchronicity.

Design/methodology/approach - Employing the special institutional background of China, this study constructs tax aggressiveness and stock price synchronicity measures for a large sample of Chinese stocks spanning the period 2003-2015. The authors employ OLS regression as the baseline methodology, and a fixed effect model, the Fama-Macbeth method and GMM as sensitivity checks. Matched samples and difference-in-difference analyses are used to control for endogeneity.

Findings – The authors find a significant and positive association between aggressive tax planning and stock price synchronicity. Because material information about risky tax transactions tends to be hidden in various tax accruals accounts, aggressive tax strategies make financial statements less transparent, thereby, increasing information asymmetry and decreasing stock price informativeness. The authors also find that the firms engaging in aggressive tax planning exhibit relatively high corporate opacity. In addition, the authors find that improvements in the tax enforcement regime, ownership status and high-quality auditors all constrain the adverse effects of tax aggressiveness.

Practical implications - This study has important practical implications for China's regulators, who are striving to reduce the tax burden of enterprises. It also helps investors to consider investment decisions more appropriately from a taxation perspective.

Originality/value - First, this paper contributes to the stock price efficiency literature by identifying the effect of a hitherto unexamined factor, namely, firm-level aggressive tax planning, on the efficiency of stock prices. Second, this study provides further empirical evidence to support the agency view of tax aggressiveness, and the informational interpretation of stock price synchronicity. Third, this study helps us better understand the effects of firm-level tax policy on firm-specific information capitalization in an environment where overall country-level investor protection is relatively weak.

Keywords China, Stock price synchronicity, Aggressive tax planning, Corporate opacity Paper type Research paper

1. Introduction

This paper investigates how aggressive tax planning affects the ability of stock prices to incorporate and reflect information in China. Hanlon and Heitzman (2010) define aggressive tax planning as the reduction of explicit taxes. Traditional theory views taxes as a cost to the firm and shareholders. Thus, transferring wealth from governments to shareholders through aggressive tax planning is viewed as a value-maximizing strategy (Phillips, 2003).

However, aggressive tax planning also generates an information transparency problem. The agency theory argument suggests that tax aggressiveness can facilitate managerial opportunism, such as earnings manipulation and resource diversion (Desai and Dharmapala, 2009). Balakrishnan et al. (2018) argue that, although tax planning provides expected tax savings, it can simultaneously increase the complexity of the organization. To the extent that this greater complexity is not adequately communicated to outside parties,

Received 3 July 2018 Revised 14 February 2019 Accepted 14 February 2019

Tax planning and stock price

synchronicity





information transparency concerns arise. We consider synchronous stock price movement stemming from tax aggressiveness as one such information transparency problem.

A substantial body of empirical literature has investigated the determinants of stock price synchronicity, predominantly from an "information transparency" perspective. Roll (1988) finds that a large proportion of stock return variation is unexplained by changes in market-wide factors, or by the announcements of value-relevant public information. As more firm-specific private information is incorporated into stock prices, the latter become less synchronous and more informative. Morck *et al.* (2000) find that stock price synchronicity is pronounced in emerging markets, owing to their relatively strong impediments to informed trading. The literature on price synchronicity, thus, finds that greater transparency and more complete revelation of firm-specific information reduces stock price synchronicity (e.g. Li *et al.*, 2004; Fernandes and Ferreira, 2008; Gul *et al.*, 2010; Kim and Shi, 2012; Hao *et al.*, 2018).

Aggressive tax planning facilitates managerial rent extraction through opaque reporting. Although many tax accruals facilitate earnings management, managers can use valuation allowances (Frank *et al.*, 2009), tax contingency reserves (Gupta *et al.*, 2016), estimates of accrued taxes (Dhaliwal *et al.*, 2004) and the designation of permanently reinvested earnings (Krull, 2004) to achieve earnings targets. Because any material information about risky tax transactions tends to be hidden in these accounts and disclosures, aggressive tax planning makes financial statements less transparent to investors and analysts, increases information asymmetry between insiders and outsiders, and decreases capital pricing efficiency.

We are motivated to study the impact of aggressive tax planning in Chinese firms, because of the uniqueness of its institutional environment. First, prior to 1979, there was no corporate income tax in China. Since the beginning of 1979, and continuing into the 1980s, the Chinese government introduced a number of enterprise taxation reforms, e.g. an income tax rate of 55 percent (35 percent) for large, state-owned enterprises (SOEs) (private firms). In 1994, China enacted the Corporate Income Tax Code, which stipulated that all domestic firms, independent of their ownership type, should pay 33 percent corporate income tax, excepting small firms with taxable income of less than 30,000 RMB, which pay 18 percent income tax. Cai and Liu (2009) argue that, although tax revenue collection has shown impressive growth since the 1994 taxation reform, both the enforcement and the collection of corporate income tax are still considered to be rather weak, engendering tax avoidance on a massive scale. In response to this, the State Administration of Taxation, China's highest tax authority, issued the Administrative Measures for the General Anti-Avoidance Rules, which went into effect on February 1, 2015. Furthermore, the 2008 major tax reform saw a uniform tax rate of 25 percent imposed on both foreign- and domestically funded businesses. This reform provides an external shock for us to test the causal relation between tax aggressiveness and its stock market consequences.

Second, China is the largest emerging economy and, now, the second largest capital market in the world. However, due to the government-dominated institutional environment (Chen *et al.*, 2017), weak corporate governance (Cheung *et al.*, 2008) and relatively poor investor protection (Gul *et al.*, 2010), the empirical findings from western countries cannot be applied easily to the Chinese context. For example, a widely documented institutional feature of China is the dominance of SOEs. On the one hand, the government is likely to charge more taxes to balance the financial budget and make adequate infrastructure investments. On the other hand, the listed SOEs are also likely to save tax and to maximize the wealth of their shareholders. The debate between these two views makes corporate tax planning a paradox for Chinese-listed firms, the incentives and the accompanying consequences for Chinese managers making firm-level tax policy may be different from those prevailing in Western countries. The institutional background in



IJMF

15.5

China allows us to gain more understanding of the interactions between different corporate governance mechanisms.

Third, and in contrast, investors in US firms (e.g.) enjoy significantly better investor protection, because of their much better-developed institutions, which ensure law/regulatory enforcement and adequate compliance via reporting, auditing and disclosure standards. Because of the generally higher level of investor protection available from governance institutions in the USA, these institutions are better able to compensate for aggressive reporting. In other words, China provides an excellent laboratory in which to study the investor protection effect of aggressive tax reporting on stock price informativeness. Finally, Chinese stocks are found to be among the least informative in the world, whereas US stocks are the most informative. Previous research attributes the lack of stock price informativeness in China to poor governance in general, and the lack of investor protection in particular (Morck *et al.*, 2000). However, the specific factors (such as aggressive tax reporting) contributing to weak investor protection in China have not been examined.

By using the market model to estimate the R^2 , and four measures of tax aggressiveness, we find that firms that engage in aggressive tax planning exhibit higher stock price synchronicity compared to firms that do not. This finding supports the agency explanation of aggressive tax planning and the information-based explanation of stock price synchronicity. Our results remain robust to alternative measures of stock price synchronicity and alternative estimation methods (i.e. firm fixed effects (FFE) specifications and generalized method of moments (GMM)). We conduct a difference-in-difference analysis in order to establish a causal relation between aggressive tax planning and stock price synchronicity. In 2008, the Chinese government carried out a taxation reform with a favorable impact on some firms, while proving costly for others. We find that firms with an increase (decrease) in tax rate, experience more (less) stock price synchronicity compared to firms with no tax rate change. We further document that the firms engaging in aggressive tax planning exhibit relatively high corporate opacity, as proxied by private corporate news hiding, abnormal discretionary accruals and the probability of informed trading measures. We also find that the positive relation between aggressive tax planning and stock price synchronicity is less pronounced in SOEs than in the non-SOEs. Finally, we find that the improvement of tax enforcement and the high-quality auditing (measured at both the firm- and individual auditor-levels) reduces price synchronicity by constraining aggressive tax strategies.

Our findings contribute to the literature in several ways. First, in recent years, a series of studies have found that the agency problem is also an important determinant or motivation for aggressive tax planning (e.g. Desai and Dharmapala, 2006; Desai et al., 2007). It can be predicted that, in countries or regions with different degrees of capital market development and legal protection levels, the economic consequences of aggressive tax planning might also be different. Therefore, it is particularly necessary to examine the economic consequences of aggressive tax planning under different capital market environments. By taking advantage of the unique Chinese settings (government dominant economics, weak investor protection and taxation reform), this paper contributes to the stock price efficiency literature by identifying the effect of a hitherto unexamined factor, namely firm-level aggressive tax planning, on the efficiency of stock prices. This provides new approaches to understanding disclosure quality from a taxation perspective, and aggregates incremental information on investor protection in emerging markets. Our findings also contribute to the ongoing debate on whether increasing tax aggressiveness provides more or less firmspecific information on the capital market (e.g. Phillips, 2003; Slemrod, 2004; Kim et al., 2011; Armstrong et al., 2012; Balakrishnan et al., 2018; Hoopes et al., 2018).

Second, Kim *et al.* (2011) document that aggressive tax planning provides the pretexts for managers to manipulate earnings, hoard corporate private news and make financial reports complex to outside investors, thus price crash risk increases. We further test this argument



Tax planning and stock price synchronicity

by applying a mediation test procedure using three proxies to measure corporate opacity directly. The results indicate a statistically significant partial mediation effect of corporate opacity on stock price synchronicity. In addition, by using the external shock of 2008 taxation reform, we provide evidence on a causal relationship between tax strategies and price synchronicity, which helps us to mitigate the endogeneity problem. Overall, our study provides more direct empirical evidence supporting the agency view of aggressive tax planning and the informational interpretation of stock price synchronicity.

The remainder of the paper is organized as follows. Section 2 provides a literature review and the hypothesis development. Section 3 presents our methodology. Section 4 includes empirical results. Section 5 is robustness tests. Section 6 concludes the paper.

2. Literature and hypothesis

Hanlon and Heitzman (2010) define aggressive tax planning as the reduction of explicit taxes: a continuum of tax planning strategies with legal tax avoidance at one end, and tax noncompliance, evasion, aggressiveness and sheltering at the other. Following Hanlon and Heitzman (2010), we do not distinguish between legal avoidance and illegal evasion, because the legality of a tax avoidance transaction is often determined *ex post* (Cai and Liu, 2009).

They propose two alternative views of aggressive tax planning. Aggressive tax strategies are usually long-term in nature, either permanently avoiding or deferring for several years the payment of taxes. As a result, the cash savings from tax avoidance can be substantial, especially relative to the modest savings that result from deferring payments for only one quarter (Mills *et al.*, 1998). Wilson (2009) estimates an average return of approximately \$12 for each \$1 in fees paid relating to tax shelters. For the risk-neutral shareholders, aggressive tax strategies, therefore, are value-maximizing activities that transfer wealth from the state to shareholders.

The alternative perspective highlights the opportunistic nature of aggressive tax strategies, from the agency theory perspective. Although aggressive tax planning provides expected tax savings, it can simultaneously increase both the opportunity for rent extraction by managers, and organizational complexity. Complex aggressive tax planning can provide management with the tools for concealing rent extraction and other resource-diverting activities (e.g. Desai and Dharmapala, 2006; Payne and Raiborn, 2018).

With respect to the stock market consequences of aggressive tax planning, Hanlon and Slemrod (2009) document a negative market reaction to tax shelter disclosure, suggesting that investors are concerned about the possibility that tax shelters are intertwined with managerial diversion and performance manipulation. Kim and Li (2014) find that stock price synchronicity is higher for offshore firms (whether having headquarters registered or subsidiaries set up), than for non-offshore firms. Furthermore, as offshore firms become more aggressive in their tax avoidance strategies, their stock prices impound increasingly less firm-specific information relative to common information. Kim *et al.* (2011) provide reliable evidence that aggressive tax planning is associated with firm-specific stock price crash risk positively, arguing that managers shield tax avoidance activities through hoarding bad news. When accumulated bad news is disclosed all at once, the stock price crashes. Kubata *et al.* (2013) use the earnings response coefficient (ERC) as a proxy for financial report informativeness and find a negative relationship between ERC and tax aggressiveness.

Our study extends the stream of research that examines the stock market consequences, proxied by stock price synchronicity, of aggressive tax planning. Stock price synchronicity refers to the synchronous movement of firm-specific stock price with the entire market (the so-called high R^2). A low R^2 means that more of the variation in stock i's price is firm-specific and asynchronous with the overall market. Thus, a low R^2 is arguably optimal, in that firm-specific variation is diversifiable.



IJMF

15.5

Cross-country determinant studies have identified that low R^2 is generally found in those countries having an advanced economic development stage (developed vs emerging markets), stronger public investor property rights (Morck *et al.*, 2000), greater capital market openness, more efficient legal systems and less corrupt economies, lower levels of inflation, smaller geographical size, better corporate governance mechanisms (Li *et al.*, 2004; Khandaker and Heaney, 2008) and individualistic cultural values (Nguyen and Truong, 2013). It is also reported that an improvement in a single country's institutional development over time is associated with a reduced R^2 (Hasan *et al.*, 2014).

At firm-level, the characteristics related to a low R^2 include high block ownership (Brockman and Yan, 2009), low level of government holding, greater foreign ownership and high audit quality (Gul *et al.*, 2010), openness to the market for corporate control (Ferreira and Laux, 2007), fewer interlocking directors on the board (Khanna and Thomas, 2009) and cross-listing status (Fernandes and Ferreira, 2008). Hutton *et al.* (2009) find that opacity, as proxied by earnings manipulation, is associated with a higher R^2 , indicating less revelation of firm-specific information. Lin *et al.* (2015) report that the high R^2 of Chinese-listed companies is explained by firms' hiding activities, possibly induced by poor quality government. Hiding activities are measured as the difference between a firm's within-industry ranking of total factor productivity and return on assets.

We posit that aggressive tax planning affects the quality of financial reporting, say through the influence of tax planning on the accrual process. Kim *et al.* (2011) argue that the differing treatments of tax planning transactions under tax and financial reporting, combined with the complexity and obfuscation of those transactions, allow managers to hide corporate private news from outside investors under the pretense of minimizing corporate tax obligations. Balakrishnan *et al.* (2018) argue that, although tax planning provides expected tax savings, it can simultaneously increase the complexity of the organization. Organizational complexity can, in turn, hinder efforts by investors to understand the firm's operations. And, to the extent that this greater complexity cannot be adequately communicated to outside parties, such as equity investors, creditors and analysts, transparency problems can arise.

There are many tax planning opportunities requiring the bifurcation of legal structures into separate business activities (e.g. income that qualifies for treaty-based withholding taxes, activities qualifying for a domestic manufacturers' deduction). We argue that these circuitous flows and separation of business activities make it more difficult for outsiders to interpret the source and persistence of the firm's earnings and cash flows and, hence, reduce the transparency of the firm's financial and operating environment, thus, increasing the R^2 [1]. Based on the preceding discussion, we develop the following hypothesis:

H1. Ceteris paribus, firms that engages in more aggressive tax planning exhibit higher stock price synchronicity, i.e. higher R^2 .

Our story is built on the speculation that aggressive tax planning provides the pretext for managers to manipulate earnings, hoard private corporate information and make the financial report complex to outside investors, thus, increasing information asymmetry and corporate opacity. As less firm-specific information is disclosed, the stock price of the tax avoidance firms moves synchronously with the market. Hutton *et al.* (2009) examine the direct effect of opaque corporate information disclosure on the R^2 . Using earnings management as a measure of opacity, they find that opacity is associated with higher R^2 , indicating less revelation of firm-specific information. Since complex tax strategies allow managers to conceal rent-extraction activities, we envision a relationship among tax planning, financial reporting opacity and R^2 . More formally, we illustrate our hypothesis for the mediation effect of corporate opacity as the following:

H2. The positive association between aggressive tax planning and high stock price synchronicity is mediated by financial reporting opacity.



Tax planning and stock price synchronicity

Unlike governments in many mature markets, the Chinese government still has absolute power over the allocation of massive state resources (e.g. licenses and permits, initial and subsequent public offerings, bank loans, subsidies and government contracts) and effective control of large-scale SOEs, which continue to dominate key sectors of the economy. On one hand, since SOEs possess substantial resources and receive policy subsidies (Li and Myers, 2006), they are required to contribute, through increased tax payments, to the support of government finance and infrastructure investment. This requirement would be expected to weaken their motivation for avoiding tax. Thus, the positive association between aggressive tax planning and stock price synchronicity is likely to be less pronounced for SOEs than for non-SOEs. On the other hand, the SOEs are listed companies and, hence, the conservation of cash through tax avoidance would help maximize shareholders' wealth and strengthen operating performance. Consequently, the incentives for avoiding tax might be equally as appealing for SOEs. Given the aforementioned competing arguments, we develop the following non-directional hypothesis:

H3. The positive association between aggressive tax planning and high stock price synchronicity varies for SOE vs non-SOE firms.

When determining aggressive tax strategies, decision makers trade off the benefits and costs of such strategies. The most obvious benefit of tax aggressiveness is greater tax savings. On the other hand, the most important cost associated with aggressive tax strategies is the possibility of being detected and sanctioned by the tax authorities (Wilson, 2009). Detection of opportunistic tax strategies requires resources, and the detection rate is likely to increase with an increase in available resources for enforcing tax regulations (Allingham and Sandmo, 1972). Prior studies document that there are significant time-series and cross-sectional differences in the tax enforcement levels among different districts (Qian and Roland, 1998; Zeng and Zhang, 2009; Fan and Tian, 2013), a factor that provides the opportunity to study the economic consequences of tax enforcement. From the perspective of corporate governance, previous literature suggests that tax enforcement results in lower tax aggressiveness (Hoopes et al., 2012), deters tax-motivated income shifting (Beuselinck et al., 2015), reduces rent diversion by managers (Desai et al., 2007), improves the quality of financial reporting (Hanlon et al., 2014) and lowers the cost of debt and implied equity capital (Kim and Zhang, 2016). Other studies based on Chinese-listed companies, also verify the existence of the above governance role of tax enforcement (Zeng and Zhang, 2009; Ye and Liu, 2011; Jiang, 2013). For example, Ye and Liu (2011) use Chinese-listed companies as a research sample and find that increasing the tax enforcement level makes it difficult for firms to manage earnings, which leads to a more transparent environment.

To summarize, tax enforcement can play the role of an external governance mechanism that will weaken the functional effect of aggressive tax planning on hiding insiders' rent extraction and hoarding bad news activities. In addition, the information asymmetry and agency costs caused by aggressive tax planning can also be weakened by increasing the tax enforcement level, thus, decreasing stock price synchronicity. We, therefore, hypothesize the following:

H4. An improvement in the tax enforcement regime reduces aggressive tax planning and, thus, stock price synchronicity.

Prior research suggests that high-quality auditing constrains aggressive tax planning (e.g. Kanagaretnam *et al.*, 2016; Richardson *et al.*, 2013). Auditors evaluate the validity of accrued taxes payable as well as tax contingent liabilities on the balance sheet, income tax expenses on the income statement, and the related note disclosures, in order to provide adequate assurance of the appropriateness of these items and disclosures to the investing public (Barrett, 2004). Although many accruals facilitate earnings management, managers can use



IJMF

15.5

valuation allowances (Frank and Rego, 2006), tax contingency reserves (Gupta *et al.*, 2016) and estimates of accrued taxes (Dhaliwal *et al.*, 2004) to achieve earnings targets. Because any material information about risky tax transactions tends to be hidden in these accounts and disclosures, auditors also have to assess whether their clients engage in potentially abusive tax transactions. High-quality auditing ensures this happens, because the ongoing interaction between the audit engagement team and tax personnel improve the auditor's understanding of a client's tax decisions, thereby, improving the quality of the tax estimates (Choudhary *et al.*, 2017). We, therefore, hypothesize the following:

H5. High-quality auditing constrains aggressive tax strategies and reduces stock price synchronicity.

3. Methodology

3.1 Measures of stock price synchronicity

To measure our dependent variable, stock price synchronicity, we first estimate the market model as follows:

$$r_{j,t} = \alpha + \beta_{1,j} \times r_{m,t-1} + \beta_{2,j} \times r_{i,t-1} + \beta_{3,j} \times r_{m,t} + \beta_{4,j} \times r_{i,t} + \beta_{5,j} \times r_{m,t+1} + \beta_{6,j} \times r_{i,t+1} + \varepsilon_{j,t},$$
(1)

where $r_{j,t}$ is the return on stock *j* in week *t*, $r_{m,t}$ is the value-weighted A-share market return in week *t*, and $r_{i,t}$ is the value-weighted industry return for industry *i* in week *t*. The value-weighted industry return is created using all firms within the same industry, with firm *j*'s weekly return omitted. In Equation (1), we also include the value of industry and market returns from t-2 to t+2 to alleviate concerns over potential non-synchronous trading bias. We require that all the return data be available for at least 180 trading days in each fiscal year. We define stock price synchronicity as the ratio of common return variation to total return variation, which is equivalent to the R^2 of the market model. To circumvent the bounded nature of R^2 within [0, 1], we use a logistic transformation of R^2 :

$$SYNCH_j = \log \left[R_j^2 / \left(1 - R_j^2 \right) \right].$$
⁽²⁾

3.2 Measures of tax aggressiveness

We use four aggressive tax planning measures (Hanlon and Heitzman, 2010). These are the effective tax rate (*ETR*), Long-term effective tax rate (*LETR*), book-tax difference (*BTD*) and *DD_BTD* (Desai and Dharmapala, 2006).

ETR is computed as in the following equation: the difference of total income tax expense minus deferred income tax expense divided by total pre-tax accounting income:

$$ETR = \frac{Total income tax expense}{Total pre tax accounting income} - \frac{Def erred income tax expense}{Total pre tax accounting income}.$$
 (3)

LRETR is computed as in the following equation: the sum of income tax paid over the previous three years divided by the sum of a firm's pre-tax income less special items:

$$LETR = \frac{\sum_{k=t-3}^{t} Cash \ tax \ paid_{i,k}}{\sum_{k=t-3}^{t} \left(Total \ pre \ tax \ accounting \ income_{i,k} - Special \ items_{i,k} \right)}.$$
 (4)

BTD is computed as follows:

للاستشارات

$$BTD = \frac{Total \ pre \ tax \ accounting \ income - (Total \ tax \ expense - Deferred \ income \ tax \ expense)/(Tax \ rate)}{Total \ pre \ tax \ accounting \ income}$$

(5)

Tax planning and stock price synchronicity IJMFThe Desai and Dharmapala (2006) residual *BTD* equals the residual from the following15,5FFE regression:

$$DD_BTD_{i,t} = \gamma_1 \times TA_{i,t} + \mu_i + \varepsilon_{i,t},\tag{6}$$

where *BTD* is the total book-tax difference, and *TA* is total accruals calculated as net income minus operating cash flows divided by lagged total asset.

3.3 Regression model

We construct our empirical model as follows:

$$SYNCH_{i,t+1} = \alpha_0 + \alpha_1 \times TAX_AGGR_{i,t} + \alpha_2 \times SIZE_{i,t} + \alpha_3 \times TURNOVER_{i,t} + \alpha_4 \\ \times STDROA_{i,t} + \alpha_5 \times VOL_{i,t} + \alpha_6 \times LEV_{i,t} + \alpha_7 \times MB_{i,t} + \alpha_8 \\ \times ROA_{i,t} + \alpha_9 \times TOP1_{i,t} + \alpha_{10} \times INSHOLD_{i,t} + \alpha_{11} \\ \times GOVHOLD_{i,t} + \alpha_{12} \times INDNUM_{i,t} + \alpha_{13} \times INDSIZE_{i,t} + \alpha_{14} \\ \times BIG4_{i,t} + \Sigma INDUSTRY + \Sigma YEAR + \varepsilon_{i,t},$$
(7)

where $SYNCH_{i,t+1}$ is stock price synchronicity in year t+1 and $TAX_AGGR_{i,t}$ is the four tax aggressiveness measures explained in Section 3.2 above. We multiply *ETR* and *LETR* by -1 so that tax avoidance increases with an increase in *ETR* and *LETR*[2]. Following previous related research (e.g. Piotroski and Roulstone, 2004; Chan and Hameed, 2006; Ferreira and Laux, 2007; Hutton *et al.*, 2009; Gul *et al.*, 2010; Balakrishnan *et al.*, 2018; Box, 2018), we include a set of control variables: firm size (*SIZE*_{*i*,*i*}), the change in average monthly share turnover (*DTURN*_{*i*,*i*}), earnings volatility (*STDROA*_{*i*,*i*}), annual trading volume turnover (*VOL*_{*i*,*i*}), leverage (*LEV*_{*i*,*i*}), market-to-book ratio (*MB*_{*i*,*i*}), return on asset (*ROA*_{*i*,*i*}), the percentage of shares held by the largest shareholder (*TOP*1_{*i*,*i*}), institutional ownership (*INSHOLD*_{*i*,*i*})[3], government ownership (*GOVHOLD*_{*i*,*i*}), the natural log of the number of firms in the industry to which a firm belongs (*INDNUM*_{*i*,*i*}), industry size (*INDSIZE*_{*i*,*i*}), and audit quality (*BIG*4_{*i*,*i*}). Finally, we controlled for the year (Σ *YEAR*) and industry fixed effects (*ΣINDUSTRY*).

4. Results

4.1 Sample selection and descriptive statistics

The data are obtained from the China Stock Market and Accounting Database and the WIND database. The initial sample contains 23,584 firm-year observations, excluding financial industry firms and "special treatment" (ST, ST*) stocks, during the period of 2003–2015. We began with 2003, because the data coverage of many of the corporate governance variables pre-2003 was rather small. Following prior studies (e.g. Gupta and Newberry, 1997; Wu *et al.*, 2012), we exclude 2,102 observations with negative pre-tax accounting income; 2,984 observations with negative *ETR* values and/or *ETR* > 1.00; 2,157 observations with less than 180 trading days per year, and 2,656 observations with missing values for the control variables, resulting in a final sample of 13,685 firm-year observations. We summarize the sample selection process in Panel A in Table I.

Panel B reports the time-series distribution of firm-year observations and the corresponding market model adjusted R^2 . The adjusted R^2 from the market model estimates range from a high of 62.8 percent in 2008 to a low of 24.9 percent in 2014. No particular time-trend is observed based on the yearly R^2 values. Panel C reports the industry distribution statistics. Firm-year observations come from a wide variety of industries with the manufacturing industry (Industry Code C) commanding the largest industry representation in our sample (61.147 percent of the observations)[4].

anel A: sample selection procedure xplanation			Observations	Tax planning and stock price
otal number of firms excludes ST, ST* and Financial industry	from 2003~2015		23,584	synchronicity
ess: ETRs with negative values or values larger than one			-2,984	
ess: Pre-tax negative accounting income ess: firm-year observations with less than 180 trading days pe			-2,102 -2,157	
ess: missing control variables	year		-2,157 -2,656	
inal sample			13,685	837
anel B: annual number of observations and annual average SYI	VCH_{t+1}			
ear Observations	% distribution		Average R^2	
003 675	4.932		0.468	
004 817	5.970		0.468	
005 795	5.809		0.464	
006 834	6.094		0.451	
803	5.868		0.416	
008 863	6.306		0.628	
009 1,028	7.512		0.476	
010 1,185	8.659		0.388	
1,388	10.142		0.454	
012 1,459	10.661		0.449	
013 1,470	10.742		0.342	
014 1,206	8.813		0.249	
015 1,162	8.491		0.541	
anel C: industry distribution of the sample observations		0/	A D ²	
ode Industry name	Observations	%	Average R^2	
Agriculture	212	1.549	0.423	
Mining	323	2.360	0.513	
Manufacturing Electricity, gas and water	8,368	61.147	0.428	
	623 368	4.552 2.689	$0.461 \\ 0.477$	
Building and construction Transportation and logistics	308 984	2.689 7.190	0.477	
Information technology	984 626	4.574	0.435	
Commerce	62	4.374 0.453	0.449	
Information, software	492	3.595	0.388	
Service	452 812	5.934	0.358	
Culture and media	162	1.184	0.423	
	36	0.263	0.429	
		0.884	0.423	
I Conglomerates	121	0.004		
I Conglomerates Water conservancy, environment and public facilities	121 47			
I Conglomerates Water conservancy, environment and public facilities		0.884 0.343 0.555	0.481 0.383	

Notes: This Table reports the sample selection and distribution. Panel A shows the steps in sample selection. Panel B shows the time-series distribution. Panel C shows the industry distribution. The Industry category is based on "guidance on the industry category of listed companies" issued by the China Securities Regulatory Commission (CSRC). Consistent with most-related studies about China, for the manufacturing industry (C), we take the two-digit code of the CSRC industry classification, for other industries, we take the one-digit code

Table I. Sample selection and distribution

Table II, Panel A presents the descriptive statistics for the sample of multivariate regression analysis. We winsorize all the continuous variables at 1 percent (99 percent) to eliminate the confounding effects of the outliers. For the dependent variable (R^2 and $SYNCH_{t+1}$), the mean and median of R^2 are 0.437 and 0.443, with a standard deviation of 0.165; The mean and median of $SYNCH_{t+1}$ are -0.293 and -0.208, with a standard deviation of 0.787. These statistics are comparable to the reported mean R^2 of 0.453 for China in the sample of Morck *et al.* (2000), and are more than double the reported mean R^2 of 0.193 for the US sample of Piotroski and Roulstone (2004). This suggests that, compared with US firms, stock prices of



IJMF	Panel A: descript	ive statist	ics								
15,5	Variables	Mean	SD	P25	Median	P75					
20,0	R_{t+1}^2	0.437	0.165	0.321	0.443	0.559					
	SYNCH _{t+1}	-0.293	0.787	-0.752	-0.208	0.256					
	ETR	0.230	0.148	0.140	0.199	0.291					
	LETR	0.225	0.112	0.150	0.205	0.287					
	BTD	0.059	0.056	0.021	0.045	0.080					
838	DD_BTD	0.086	0.035	0.067	0.083	0.101					
	SIZE	21.936	1.217	21.082	21.771	22.623					
	DTURN	-0.026	0.380	-0.174	0.009	0.176					
	STDROA	0.022	0.036	0.006	0.012	0.023					
	VOL	5.119	3.606	2.393	4.165	6.859					
	LEV	0.470	0.197	0.325	0.477	0.619					
	MB	3.253	2.694	1.621	2.464	3.935					
	ROA	0.049	0.040	0.020	0.040	0.067					
	ACCM	0.179	0.136	0.084	0.143	0.234					
	HIDE	0.296	0.213	0.120	0.254	0.439					
	PIN	0.197	0.132	0.074	0.168	0.298					
	TOP1	0.380	0.158	0.252	0.363	0.500					
	INSHOLD	0.340	0.251	0.105	0.321	0.543					
	GOVHOLD	0.149	0.226	0.000	0.000	0.299					
	MANHOLD	0.036	0.111	0.000	0.000	0.000					
	INDNUM	5.291	1.138	4.394	5.509	6.163					
	INDSIZE	27.828	1.368	27.017	27.936	28.837					
	BIG4	0.067	0.251	0.000	0.000	0.000					
	Panel B: correlat										
	1 D ²	1	2	3	4	5	6	7	8	9	10
	1. R_{t+1}^2	-									
	2. $SYNCH_{t+1}$	0.21	-								
	3. <i>ETR</i> *	0.07	0.06	-							
	4. <i>LETR</i> *	0.07	0.05	0.77	-						
	5. <i>BTD</i>	0.15	0.05	0.28	0.24	-					
	6. <i>DD_BTD</i>	0.05	0.01	0.06	0.07	0.20	-				
	7. SIZE	0.11	0.15	0.07	0.07	0.02	-0.01	-			
	8. DTURN	-0.08	0.05	0.06	0.07	-0.07	-0.10	0.06	-		
	9. STDROA	-0.06	-0.04	-0.07	-0.06	0.24	0.10	-0.14	0.00	-	
	10. VOL	-0.04	0.09	0.00	0.03	-0.05	0.01	-0.21	0.34	0.05	-
	11. <i>LEV</i>	0.05	0.07	0.20	0.25	-0.28	-0.02	0.38	0.09	0.05	-0.05
	12. <i>MB</i>	-0.27	-0.06	-0.07	-0.07	0.29	0.08	-0.26	0.08	0.22	0.27
	13. <i>ROA</i>	-0.13	-0.05	-0.29	-0.26	0.89	0.15	-0.03	-0.08	0.18	-0.06
	14. TOP1	0.04	0.03	-0.01	0.00	0.09	-0.02	0.22	-0.06	-0.04	-0.15
	15. INSHOLD	-0.13	-0.10	-0.05	-0.05	0.21	0.02	0.42	-0.08	-0.06	-0.29
	16. GOVHOLD	0.11	0.13	0.03	0.04	0.01	-0.10	0.02	0.06	-0.02	-0.02
	17. MANHOLD	-0.13	-0.15	-0.08	-0.12	0.07	0.09	-0.12	-0.15	-0.03	0.14
	18. INDNUM	-0.10	-0.10	-0.11	-0.15	0.02	0.08	-0.02	-0.04	0.01	0.06
	19. INDSIZE	-0.08	-0.07	-0.06	-0.09	0.03	0.09	0.24	-0.04	0.00	0.06
	20. BIG4	0.04	0.04	-0.01	-0.02	0.04	-0.05	0.37	0.01	-0.03	-0.13
	0	11	12	13	14	15	16	17	18	19	20
	1. R_{t+1}^2										

1. *K*_{t+1} 2. *SYNCH*_{t+1} 3. *ETR** 4. *LETR** 5. *BTD* 6. *DD_BTD*

المنسارات

Table II.Descriptive statisticsand correlation matrix

(continued)

7. SIZE 8. DTURN 9. STDROA 10. VOL											Tax planning and stock price synchronicity
11. <i>LEV</i>	-										
12. <i>MB</i>	-0.04	-									
13. ROA	-0.38	0.28	-								
14. TOP1	0.01	-0.07	0.08	-							839
15. INSHOLD	0.07	0.13	0.22	0.16	-						
16. GOVHOLD	0.04	-0.10	-0.02	0.39	-0.31	-					
17. MANHOLD	-0.24	0.05	0.09	-0.09	-0.15	-0.21	-				
18. INDNUM	-0.10	0.03	0.02	-0.09	0.07	-0.23	0.19	-			
19. INDSIZE	0.00	0.02	0.02	-0.02	0.27	-0.29	0.19	0.79	-		
20. BIG4	0.04	-0.11	0.06	0.13	0.14	0.06	-0.07	-0.04	0.03	-	

Notes: This table presents the summary statistics for the variables used in our regression model. The data were obtained from the WIND and CSMAR databases for the period 2003–2015. The final samples consisted of 13,685 observations. We winsorize all the continuous variables at 1 percent (99 percent) to eliminate outlier effects on the regression results. Variable definitions are in Table AI; This panel provides Pearson correlation coefficients for the key variables used in the regression analyses. Italic coefficients are significant at 5 percent or better

Table II.

Chinese-listed firms tend to co-move, to a greater (less) extent, with market-wide and/or industry-wide information (firm-specific information).

The mean and median of *ETR* are 0.230 and 0.199, with a standard deviation of 0.148; the mean and median of *LETR* are 0.225 and 0.205, with a standard deviation of 0.112 (recall that we multiplied these measures by -1 for regression purposes, but report the untransformed values in the descriptive statistic section); the mean and median of *BTD* are 0.059 and 0.045, with a standard deviation of 0.056; and the mean and median of *DD_BTD* are 0.086 and 0.083, with a standard deviation of 0.035.

Panel B presents the Pearson correlations between our major variables. Three of the four measures of aggressive tax planning (*ETR**, *LETR**, and *BTD*) are correlated significantly and positively with *SYNCH*_{t+1} (at p < 0.05 and better). *SYNCH*_{t+1} is also correlated positively with firm size (p < 0.01), turnover, volatility, leverage and some corporate governance measures.

4.2 Baseline regression

We present our baseline regression results in Table III. We predict a positive association between aggressive tax strategies and stock price synchronicity, which will support H1. To alleviate concern about potential cross-sectional and time-series dependence in the data, we report *t*-values on an adjusted basis, using robust standard errors corrected for double (firm and year) clustering (Gow *et al.*, 2010; Petersen, 2009).

Column (1) reveals that the coefficient on ETR^* is positive and marginally significant (p < 0.10). The reported coefficient of 0.082 suggests that a one standard deviation increase in ETR^* is associated with a 1.22 percent increase in $SYNCH_{t+1}$ (0.0824×0.148). The coefficients on the other three tax proxies are also positive and statistically significant (coefficients of 0.16, 1.12 and 0.48 for the $LETR^*$, BTD and DD_BTD measures, respectively, all significant at p < 0.01). We, therefore, find support for H1. The coefficients of the control variables are generally consistent with the findings of prior studies. Consistent with Gul *et al.* (2010), we find that the coefficients on SIZE, *INDSIZE* are significantly positive. We also find that *MB*, *LEV*, *VOL*, *ROA*, *INSHOLD*, *GOVHOLD*, *MANHOLD* and *BIG4* are all negatively related to stock price synchronicity.



IJMF			SVM	CH_{t+1}	
15,5	DV	Column (1)	Column (2)	Column (3)	Column (4)
	ETR*	0.0821* (1.94)	_	_	_
	LETR*	- ` ` `	0.159*** (2.64)	-	_
	BTD	-	- , , ,	1.120*** (5.50)	-
0.40	DD_BTD	-	-	- ,	0.477*** (2.79)
840	SIZE	0.144*** (15.35)	0.143*** (15.32)	0.147*** (15.63)	0.143*** (15.26)
	DTURN	0.0123 (0.67)	0.0127 (0.69)	0.0103 (0.56)	0.0143 (0.77)
	STDROA	0.137 (0.77)	0.125 (0.70)	0.271 (1.47)	0.112 (0.63)
	VOL	-0.0507 ** (-2.01)	-0.0511 ** (-2.02)	-0.0524 ** (-2.07)	-0.0501 ** (-1.99)
	LEV	-0.294 *** (-6.75)	$-0.286^{***}(-6.55)$	$-0.287^{***}(-6.56)$	$-0.296^{***}(-6.80)$
	MB	-0.0254^{***} (-8.02)	-0.0254^{***} (-8.06)	$-0.0243^{***}(-7.76)$	-0.0256^{***} (-8.08)
	ROA	-0.802^{***} (-3.88)	$-0.811^{***}(-3.98)$	$-0.689^{**}(-2.18)$	$-0.792^{***}(-3.93)$
	TOP1	0.0243 (0.52)	0.0254 (0.54)	0.0233 (0.50)	0.0226 (0.48)
	INSHOLD	$-0.350^{***}(-9.08)$	-0.351 * * * (-9.12)	$-0.349^{***}(-9.04)$	-0.344^{***} (-8.94)
	GOVHOLD	$-0.156^{***}(-4.52)$	$-0.155^{***}(-4.50)$	-0.147^{***} (-4.28)	$-0.152^{***}(-4.42)$
	MANHOLD	-0.509^{***} (-8.13)	-0.513*** (-8.19)	-0.504^{***} (-8.07)	-0.511*** (-8.17)
	INDNUM	-0.0205(-0.45)	-0.0217 (-0.48)	-0.0167 (-0.37)	-0.0177 (-0.39)
	INDSIZE	0.0436* (1.69)	0.0448* (1.73)	0.0416 (1.61)	0.0412 (1.59)
	BIG4	-0.106^{***} (-3.45)	$-0.106^{***}(-3.46)$	-0.112*** (-3.64)	-0.102*** (-3.33)
	INDUSRY& YEAR	Yes	Yes	Yes	Yes
	INTERCEPT	-4.028^{***} (-7.02)	-4.116^{***} (-7.12)	-5.135*** (-8.44)	-3.932*** (-6.87)
	Observation	13,685	13,685	13,685	13,685
	F-statistics	186.35***	186.56***	186.27***	187.49***
	Adj. R ²	0.3924	0.3926	0.3934	0.3926
	stock price synchron	icity. The dependent	t of the relation betwee variable is $SYNCH_{t+}$	1. The independent v	ariables are the four

Table III. Baseline regression result

4.3 The mediation test: Does aggressive tax planning induce corporate opacity?

the 10, 5 and 1 percent levels, respectively (two-sides)

The story of this paper is consistent with the following view: aggressive tax planning provides the pretexts for managers to manipulate earnings, hoard corporate private news and make financial reports complex to outside investors, all of which increase information asymmetry and corporate opacity. As a result, less firm-specific information is incorporated into stock prices, thus, price synchronicity is increased. We test this prediction by applying a mediation test procedure and using three proxies to measure corporate opacity, namely, three years' accumulated abnormal accruals (*IACCM*) (Hutton *et al.*, 2009), degree of private corporate news hiding (*HIDE*) (Lin *et al.*, 2015) and, finally, the probability of informed trading (*PIN*) (Easley and O'Hara, 2004).

proxies of aggressive tax planning: *ETR**, *LETR**, *BTD*, *DD_BTD*, respectively. We use the OLS model to estimate the regression. To alleviate the concern about potential cross-sectional and time-series dependence in the data, we report *t*-values on an adjusted basis, using robust standard errors corrected for double (firm and

year) clustering (Gow et al., 2010; Petersen, 2009). Variable definitions are in Table AI. *,**,***Significant at

We follow the mediation test approach of Baron and Kenny (1986) who propose that a mediation effect exists when the following three conditions are fulfilled. Path A: variations in the levels of the independent variable (i.e. *ETR**, *LETR**, *BTD* and *DD_BTD*) account significantly for variations in the proposed mediators (i.e. *ACCM*, *HIDE*, or *PIN*). Path B: variations in the proposed mediators account significantly for variations in the dependent variable (*SYNCH*_{t+1} in our study). Path C: the significant relationship between aggressive tax strategies (i.e. *ETR**, *LETR**, *BTD* and *DD_BTD*) and *SYNCH*_{t+1} (the baseline regression) becomes insignificant once Paths A and B are controlled (full mediation); or the significant relation between *TAX_AGGR* and *SYNCH*_{t+1} in the baseline model is reduced once Paths A and B are controlled (partial mediation).



Executing this test requires three separate regressions in addition to the baseline Tax planning and stock price

 $MV_{i,t} = \alpha_0 + \alpha_1$ synchronicity $\times TAX_AGGR_{i,t} + \sum CONTROLS + \sum INDUSTRY + \sum YEAR + \varepsilon_{i,t},$ 8(a) 841

 $SYNCH_{i,t+1} = \alpha_0 + \alpha_1$ $\times MV_{i,t} + \sum CONTROLS + \sum INDUSTRY + \sum YEAR + \varepsilon_{i,t}, 8(b)$

$$SYNCH_{i,t+1} = \alpha_0 + \alpha_1 \times TAX_AGGR_{i,t} + \alpha_2$$
$$\times MV_{i,t} + \sum CONTROLS + \sum INDUSTRY + \sum YEAR + \varepsilon_{i,t}, \ 8(c)$$

where *MV* are the proposed mediators, i.e. *IACCM*, *HIDE* or *PIN*. A detailed explanation of the estimation procedures for these variables is reported in Table AI. Results are presented in Table IV. Panel A, Model 2, reveals a positive and significant association between the *IACCM* and *SYNCH*_{*t*+1}, suggesting that opaque financial information increases stock price synchronicity (e.g. coefficient on *IACCM* is 0.2042 (p < 0.01)). Importantly, the coefficients on the aggressive tax planning measures remain positive and significant after controlling *IACCM* (coefficients on *ETR**, *LETR**, *BTD*, *DD_BTD* are 0.0834 (p < 0.05), 0.159 (p < 0.01), 1.2082 (p < 0.01) and 0.6162 (p < 0.01), respectively). The Sobel *z*-statistics are significant across all the aggressive tax planning measures (significant at p < 0.05 and better). Overall, the tabulated results indicate a statistically significant partial mediation effect of *ACCM* on stock price synchronicity.

Panel B, Model 2, reveals a positive and significant association between the *HIDE* and *SYNCH*_{*t*+1}, suggesting that hoarding bad news increases stock price synchronicity (coefficient on *HIDE* is 0.0318, *t*-statistic 2.09, (p < 0.05)). Moreover, the coefficients on the aggressive tax planning measures remain positive and significant after controlling *HIDE* (coefficients on *ETR**, *LETR**, *BTD*, *DD_BTD* are 0.0815 (p < 0.10), 0.1414 (p < 0.05), 1.1507 (p < 0.01), and 0.6817 (p < 0.01), respectively). The Sobel *z*-statistics are significant across all the aggressive tax planning measures (significant at p < 0.01). The tabulated results indicate a statistically significant partial mediation effect of *HIDE* on stock price synchronicity.

Finally, Panel C provides test results for the mediation effect of *PIN*, which is qualitatively consistent with the other two measures (*ACCM*, *HIDE*) of corporate opacity (coefficient on *ETR**, *LETR**, *BTD*, *DD_BTD* are 0.0801 (p > 0.10), 0.1389 (p < 0.05), 1.078 (p < 0.01), and 0.6542 (p < 0.01), respectively). Overall, the results in Table VIII indicate a statistically significant partial mediation effect of corporate opacity on stock price synchronicity.

4.4 SOEs, tax aggressiveness and price synchronicity

To test *H3*, we run Equation (7) for SOEs (if the ultimate controlling owner is the central/local government or a collective). Panel A of Table V reports the univariate test of difference in mean tax aggressiveness measures between the SOEs and the non-SOEs. Across all four proxies, we document that tax aggressiveness is weaker for SOEs than for their non-SOE counterparts (e.g. the mean value of ETR^* is -0.256 (-0.188) for the SOE (non-SOE) groups with the difference being significant at p < 0.10).

IJMF 15,5	Column (8)	IACCM	$\begin{array}{cccccc} 0.625^{***} & (9.40) & 0.637^{***} & (10.13) \\ Yes & Yes & Yes \\ 13,685 & 13,685 \\ 0.2613 & 0.2704 \end{array}$		HIDE	N.150 (1.56) 0.201**** (2.73) Yes Yes Yes Yes 13,591 13,591 0.03386 0.0392	(continued)
842	Column (7)	IACCM	0.625*** (9.40) Yes Yes 13.685 0.2613		HIDE	0.150 (1.56) Yes Yes 13,591 0.0386	
	Column (6)	IACCMI	0.0210*** (2.32) Yes 13,685 0.2495	~~~~~	HIDE).0829*** (2.65) Yes 13,591 0.0398	
	Column (5)	IACCM	0.0223** (2.36) Yes 13.685 0.2498	/ 0.2042*** (3.15) Yes Yes 0.3924 0.3924	HIDE		
	Column (4)	DD_BTD	0.477*** (2.79) Yes 13.685 0.3926	$\begin{array}{cccc} 0.6162^{****} & (3.50) & / \\ 0.2293^{**} & (2.48) & 0.2042^{****} & (3.15) \\ Yes & Yes & Yes \\ 13.685 & 0.3924 & 0.3924 \\ 0.7931^{**} & - \\ & (0.05) & - \end{array}$	DD_BTD	0.6081*** (3.42) Yes Yes 13,591 0.3896	
	Column (3)	BTD	1.120*** (5.50) Yes 13,685 0.3934	1.2082**** (5.90) 0.2409**** (3.71) Yes Yes 13.685 0.3937 0.7823****	BTD	$ \begin{array}{ccccccc} 0.1414^{**} & (2.31) & 1.1551^{***} & (5.68) & 0.6081^{***} & (3.42) \\ Yes & Yes & Yes \\ Yes & Yes & Yes \\ 13,591 & 13,591 & 13,591 \\ 0.3894 & 0.3903 & 0.3896 \\ \end{array} $	
	Column (2)	LETR*	0.159*** (2.64) Yes 13,685 0.3926	0.159**** (2.65) 0.2040**** (3.15) Yes Yes 13.685 0.3928 0.7761**** (0.05)	LETR*	0.1414** (2.31) Yes Yes 13,591 0.3894	
	Column (1) diator	lator) ETR*	0.082* (1.94) Yes Yes 13,685 0.3924 m)	<pre>.00834** (1.97) 2050**** (3.16) Yes Yes 13.685 0.3926 0.37732** (0.05)</pre>	<i>ator</i> iator) <i>ETR</i> *	0.0775* (1.82) Yes 13,591 0.3892	
Table IV. The mediation effect of corporate opacity on the relationship between aggressive tax planning and synchronicity	Colt Panel A: IACCM as the mediator	Model (1) (without the mediator) DV (1~4): $SYNCH_{t+1}$; DV	Definition of the second secon	DV $(1 \sim b)$: STNCH ₁₊₁ TAX_AGGR OTHER CONTRROLS INDUSRY & YEAR INTERCEPT Adj: R^2 Sobel Z (p-value) of Sobel Z	Panel B: HIDE as the mediator Model (1) (without the mediator) $DV (1 \sim 4)$: SYNCH ₄₊₁ : DV	10-91. ILLE TAX_AGGR TAX_AGGR INDUSRY & YEAR Observation Adj. R ²	
للاستشارات	äj		المن				

WWW.

	PIN* \$*** (2.66) Yes Yes 13.685	0.2632 / / / / / / / / / / / / / / / / / / /	Tax planning and stock price synchronicity
	PIN* PIN* 0.927*** (3.19) 0.618*** (2.66) Yes Yes Yes 13.685 13.685	0.2633 0.2633	843
	PIN* PIN* 0.189** (2.45) Yes Yes 13.685	0.2632 ///////////////////////////////////	
/ 0.0318** (2.09) Yes 13,591 0.3891	<i>PIN</i> * 0.0956* (1.68) Yes 13.685	0.2630 / .0518*** (2.79) Yes Yes 13,685 13,685 13,685 13,685 allevelter the conc crited for double (fi screet for double (five, fi	
0.6817*** (3.39) 0.0292** (2.00) Yes 13,591 0.3897 0.3213***	0.01 DD_BTD 0.477*** (2.79) Yes Yes 13.685	0.3926 0.2630 0.6542*** (2.69) / 0.04672** (2.24) 0.0518*** (2.79) Yes Yes Yes 13,685 0.3956 0.3956 0.3931 0.16 onship between aggressive tax pla 0.16 0.30 <i>BTD</i> , respectively. The de lin our models. To alleviate the co andard errors corrected for double 1 nercent levels. respectively (rouble	
1.1507*** (5.66) (0.0289** (5.66) (0.0289** (2.01) Yes 13,591 0.3903 0.3215***	0.01 BTD 1.120*** (5.50) Yes Yes 13.685	0.3934 1.078*** (3.27) (0.0513** (3.26) (Yes Yes 13,685 0.3955 0.3955 0.3955 0.3955 0.3824 0.12 city on the relatio city on the relatio R*, <i>LETR*</i> , <i>BTL</i> ects are included at the 10,5 and 1 at the 10,5 and 1	
0.1414** (2.36) 0.0358** (2.18) Yes 13.591 0.3894 0.3894 0.322***	0.01 <i>LETR*</i> 0.159*** (2.64) Yes Yes 13.685	0.3926 0.1389** (2.02) 0.0427*** (2.58) Yes Yes 13,685 0.3955 0.4279** 0.05 ct of corporate opa tax planning: <i>ET</i> industry fixed effi in an adjusted basis i an an adjusted basis	
r) 0.0815* (1.91) 0.0356** (2.21) Yes 13,591 0.3893 0.3893		0.3924 r) 0.0801 (1.61) 0.0476** (2.26) Yes Yes 13,685 0.3954 0.4282?*** 0.4282?*** 0.01 0.01 he mediation effetees of aggressive tes of aggressive	
Model (2) (with the mediator) DV (1~5): SYNCH ₄₊₁ TAX_AGGR HIDE OTHER CONTRROLS NUUSRY & YEAR Observation Adj. R ² Sobel Z	Panel C: PIN as the mediator Model (1) (without the mediator) $DV (1 \sim 4)$: SYNCH ₍₊₁): DV (5-8): PIN (5-8): PIN $TAX_AGGR 0.0$ OTHER CONTRROLS INDUSRY & YEAR Observation	Adj. R^2 0.3924 0.3926 0.3924 0.3926 0.3926 0.3926 0.2630 0.2633 0.	Table IV.
مدار الم للاستشارات		≥ದ Z ⊱ ರ ರ ನ	ww

IJMF 15,5	Panel A: univa	riate test of d ET			ressiveness in FR*		n-SOE firms FD	DD	BTD
		Column (1)	Column (2)	Column (3)	Column (4)	Column (5)	Column (6)	Column (7)	
		SOE	NON-SOE	SOE	NON-SOE	SOE	NON-SOE	SOE	NON-SOE
	Mean	-0.256	-0.188	-0.247	-0.203	0.045	0.063	0.061	0.093
	Observation	8,895	4,790	8,895	4,790	8,895	4,790	8,895	4,790
	Diff.	0.0	52*	0.0	71*	0.00	6***	0.03	31**
844	Panel B: regres	sion results f	or SOE and	non-SOE firn	ns				
	•	Column (1) SOE	Column (2) NON-SOE	Column (3) SOE	Column (4) NON-SOE	Column (5) SOE	Column (6) NON-SOE	Column (7) SOE	Column (8) NON-SOE
	ETR*	0.053** (2.03)	0.107* (1.85)	-	-	-	-	-	-
	LETR*	-	_	0.067** (2.35)	0.073*** (2.10)	-	-	-	-
	BTD	-	-	-	-	1.007*** (4.37)	1.526** (2.11)	-	-
	DD_BTD	-	_	-	_	(4.57)	-	0.254** (2.03)	0.365* (1.91)
	OTHER	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
	CONTRROLS INDUSRY &	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
	YEAR INTERCEPT	-6.133***	-4.156	-3.973***	-3.865***	-4.497***	-3.525***	-2.756***	-1.987***
	IIVI LICOLI I	(-5.13)	(-3.27)	(-6.12)	(-5.54)	(-6.31)	(-4.03)	(-3.97)	(-3.11)
	Observation	8,895	4,790	8,895	4,790	8,895	4,790	8,895	4,790
	F-statistics	98.29***	79.52***	96.34***	82.45***	97.26***	83.23***	99.03***	84.52***
	Adj. R^2	0.3459	0.3013	0.3461	0.3023	0.3445	0.3009	0.3451	0.3007
	Hausman test	0.0	57*	0.2	265	0.00	6***	0.03	31**
	Notes: This ta	ble presents t	he associatio	n between as	gressive tax	planning and	l svnchronici	tv for SOEs y	vs non-SOEs.
	The independe								
	respectively. T			*	00	*			
	alleviate the co								
Table V.	adjusted basis,					*		*	
SOEs vs non-SOEs	2009). Variable	0			(/ 0.	,	, , ,

Panel B reports the regression results of Equation (8) for SOE and non-SOE groups. Regression results for both sub-groups reveal that aggressive tax planning increases price synchronicity. Importantly, the Hausman test indicates that the coefficients of the four tax proxies in the SOE group are lower those in the non-SOE group (e.g. the coefficients of *ETR** are 0.053 for the SOE group, but 0.107 for the non-SOE group. The difference is significant at p < 0.10). The results for the remaining tax aggressiveness measures are generally consistent.

Overall, the results in Table V support the view that the motivation for aggressive tax planning is weaker for SOEs than for non-SOE firms.

4.5 The moderating effects of tax enforcement

Following Mertens (2003), we measure tax enforcement using the following equation to test H4:

$$T_{i,t}/GDP_{i,t} = \alpha_0 + \alpha_1 \times IND1_{i,t}/GDP_{i,t} + \alpha_2 \\ \times IND2_{i,t}/GDP_{i,t} + \alpha_3 + TRADE_{i,t}/GDP_{i,t} + \varepsilon_{i,t},$$
(9)

where $T_{i,t}$ is tax revenue of the provincial government at the end of the fiscal year; $GDP_{i,t}$ is the gross domestic product of the provinces at the end of fiscal year; $IND1_{i,t}$ is the first industry output; $IND2_{i,t}$ is the second industry output; $TRADE_{i,t}$ is the total volume of

foreign trade. We divide actual $T_{i,t}/GDP_{i,t}$ by the predicted $T_{i,t}/GDP_{i,t}$ derived from estimating Equation (9) and denote this as $FORCE_{i,t}$. The greater the value of $FORCE_{i,t}$, the and stock price higher the level of tax enforcement. We then interact our respective tax measures with $FORCE_{i,t}$ and include the interactive variables in our baseline regression model. We expect the interactive coefficients to be negative and significant, to support the notion that better tax enforcement reduces aggressive tax strategies and, therefore, price synchronicity. Regression results are presented in Table VI. Three of the four interactive variables are negative and significant (coefficients -0.0835 (p < 0.01), -0.0831 (p < 0.01) and -0.563(p < 0.01) for ETR*, LETR* and DD BTD tax proxies, respectively). Therefore, we conclude that a high level of tax enforcement attenuates the adverse effects of aggressive tax strategies on stock price informativeness.

4.6 The moderating effects of high-quality auditing

We now provide test results for H5, where we hypothesized that high-quality auditing can mitigate aggressive tax strategies and, hence, can increase stock price informativeness by reducing price synchronicity. We use both firm-level as well as individual auditor-level audit quality indicators.

At the firm level, we use audit firm industry specialization as the audit quality indicator. We follow Minutti-Meza (2013), and measure audit firm specialization using market shares. In Equation (10), $MSA_FIRM_{m,k,t}$ measures the market share for each industry-year, based on client's size (i.e. assets) by industry k. N represents the number of clients that are served by audit firm m in industry k and M is the number of audit firms in industry k. We calculate $MSA_FIRM_{m,k,t}$ for every audit firm by year and industry. If $MSA_FIRM_{m,k,t}$ is above the sample median (across year), we take the value of

DV: SYNCH _{t+1}	Column (1) ETR*	Column (2) LETR*	Column (3) BTD	Column (4) DD_BTD
TAX_AGGR	0.0373 (0.83)	0.111* (1.78)	0.547 (0.93)	1.023 (1.55)
TAX_AGGR*FORCE	-0.0835*** (-3.16)	-0.0831*** (-3.05)	-0.549 (-0.99)	-0.563*** (-2.85)
FORCE	-0.0769(-1.50)	-0.0792(-1.51)	-0.473 (-0.89)	-0.0931(-1.45)
OTHER CONTRROLS	Yes	Yes	Yes	Yes
INDUSRY & YEAR	Yes	Yes	Yes	Yes
INTERCEPT	$-3.874^{***}(-6.73)$	$-3.959^{***}(-6.83)$	$-4.569^{***}(-5.43)$	$-4.012^{***}(-6.99)$
Observation	13,685	13,685	13,685	13,685
F-statistics	180.85***	180.77***	179.37***	180.28***
Adj. R ²	0.3931	0.3933	0.3936	0.3927

Notes: This table presents the test of the moderating effect of tax enforcement on the relation between aggressive tax planning and stock price synchronicity. The independent variables are the four proxies of aggressive tax planning: ETR*, LETR*, BTD, DD_BTD, respectively. The dependent variable is SYNCH in year *t*+1. The moderating variable is *FORCE*, which is estimated from the following model: $T_{i,t}/GDP_{i,t} = \alpha_0 + \frac{1}{2}$ $\alpha_1 \times IND1_{i,l'}(GDP_{i,t} + \alpha_2 \times IND2_{i,l'}(GDP_{i,t} + \alpha_3 \times TRADE_{i,l'}(GDP_{i,t} + \varepsilon_{i,t})$ where $T_{i,t}$ is the tax revenue of local government at the end of a fiscal year (province level); $GDP_{i,t}$ is the gross domestic product of local government at the end of a fiscal year (province level); IND1_{i,t} is the first industry output; IND2_{i,t} is the second industry output; TRADE_{i,t} is the total volume of foreign trade. We run a cross-sectional regression and get the predicted $T_{i,t}/GDP_{i,t}$, then we divide real $T_{i,t}/GDP_{i,t}$ by predicted $T_{i,t}/GDP_{i,t}$ to obtain $FORCE_{i,t}$. The greater the FORCE_{i,b} the higher the level of tax enforcement. Year and industry fixed effects are included in our models. To alleviate the concern about potential cross-sectional and time-series dependence in the data, we report t-values on an adjusted basis, using robust standard errors corrected for double (firm and year) clustering (Gow et al., 2010; Petersen, 2009). Variable definitions are in Table AI. *, *** Significant at the 10 and 1 percent levels, respectively (two-sides)

Table VI. Tax enforcement regime, aggressive tax planning and price synchronicity

Tax planning synchronicity

EXPERT_FIRM as one, and zero otherwise. We use *EXPRT_FIRM* as the proxy of auditor firm industry specialization:

$$MSA_FIRM_{m,k,t} = \sum_{n=1}^{N} \sqrt{ASSET_{m,k,n,t}} / \sum_{m=1}^{M} \sum_{n=1}^{N} \sqrt{ASSET_{m,k,n,t}}.$$
 (10)

At the individual auditor level analysis, we use individual auditor industry specialization as a proxy for high-quality auditing. Equation (11) below is used to measure individual-level industry specialization. $MSA_AUDITOR_{i,k,t}$ measures the market share for each industry-year based on client's size for individual auditor *i* by industry *k* in fiscal year *t*. *J* represents the number of clients that are served by individual auditor *i* in industry *k* and fiscal year *t*. Usually, each firm has two auditors, so we take the average value of $MSA_AUDITOR_{i,k,t}$. As in the case of $EXPERT_FIRM$, if the average value of $MSA_AUDITOR_{i,k,t}$ of the auditors for each firm is above the sample median across year, we value $EXPRT_AUDITOR$ as one, and zero otherwise. We use $EXPRT_AUDITOR$ as the proxy of individual auditor level industry specialization:

$$MSA_AUDITOR_{i,k,t} = \sum_{j=1}^{J} \sqrt{ASSET_{i,k,t,j}} / \sum_{i=1}^{I} \sum_{j=1}^{J} \sqrt{ASSET_{i,j,k,t}}.$$
 (11)

We include various audit quality indicators and interact those with the aggressive tax measures, to infer whether high-quality auditing reduces price synchronicity by mitigating aggressive tax strategies. Table VII reports the regression results. Panel A reveals that the coefficients on the interactive variable (*TAX_AGGR*×*EXPERT_FIRM*) are negative and significant, suggesting that audit firm industry specialization attenuates aggressive tax strategies and, hence, makes stock prices more informative. Panel B shows that individual auditor-level industry specialization also reduces aggressive tax strategies and price synchronicity above and beyond the audit firm specialization effects (e.g. the interactive coefficients are -0.07 (p < 0.10); -0.22 (p < 0.05); -0.11 (p < 0.05) and -0.60 (p < 0.10) for the *ETR**, *LETR**, *BTD* and *DD_BTD* measures, respectively). Overall, we conclude that high-quality auditing measured at both the firm and individual auditor level improves stock price informativeness by constraining aggressive tax strategies.

5. Robustness

IJMF

15.5

846

5.1 Alternative measures of stock price synchronicity

The institutional features of Chinese markets require us to use three different specifications of the market model from which we derive two alternative measures of synchronicity. In China, some firms issue A-shares only, some issue both A+B shares, while some other firms issue A+H shares. Our primary *SYNCH* measure in Equation (1) did not consider this specification. In fact, returns on stocks of A+B (A+H) share firms are likely to co-move with B-share (H-share) market factors, in addition to A-share market factors. To address this issue, we also estimate, for each fiscal year, the market model for firms with only domestic A-shares, using Equation (12a), and estimate two different market models for firms with A+B shares and firms with A+H shares using Equations (12B) and (12C), respectively:

$$r_{j,t} = \alpha + \beta_{1,j} \times r_{m,t-1} + \beta_{2,j} \times r_{i,t-1} + \beta_{3,j} \times r_{m,t} + \beta_{4,j} \times r_{i,t} + \beta_{5,j} \times r_{m,t+1} + \beta_{6,j}$$



(12a)

DV: SYNCH _{t+1}	Column (1) ETR*	Column (2) LETR*	Column (3) BTD	Column (4) DD_BTD	Tax planning and stock price synchronicity
Panel A: firm-level analy	sis				
TAX_AGGR	0.0156 (0.30)	0.0446 (0.61)	0.997*** (4.34)	0.485** (2.06)	
TAX_AGGR×					947
EXPERT_FIRM	-0.144* (-1.93)	· · · ·	$-0.0802^{**}(-2.37)$	-0.0164** (-2.05)	847
EXPERT_FIRM	$-0.149^{**}(-2.52)$			-0.0387** (-2.25)	
OTHER CONTRROLS INDUSRY & YEAR	Yes Yes	Yes Yes	Yes Yes	Yes Yes	
INDUSKI & ILAK INTERCEPT	-3.925*** (-6.81)	-3.988*** (-6.87)	-4.955^{***} (-7.98)	-3.882*** (-6.74)	
Observation	13,685	13,685	13,685	13,685	
F-statistics	176.85	176.95	177.19	178.95	
Adj. R^2	0.3910	0.3914	0.3916	0.3911	
Panel B: individual audi	tor level: (Include EXI	PRT FIRM)			
TAX_AGGR	0.0485 (0.87)	0.0589 (0.75)	0.912*** (3.82)	0.138 (0.55)	
$TAX_AGGR \times$					
EXPERT_AUDITOR	-0.0696* (-1.92)	· · · ·	()		
EXPERT_AUDITOR	$-0.0591^{**}(-2.10)$		-0.0581 ** (-2.26)		
EXPERT_FIRM	(/	-0.0387*** (-3.07)	()		
OTHER CONTRROLS	Yes	Yes	Yes	Yes	
INDUSRY & YEAR INTERCEPT	Yes	Yes	Yes	Yes	
Observation	-3.997*** (-6.94) 12,841	$-4.055^{***}(-7.01)$ 12,841	-4.908*** (-7.83) 12,841	-3.859^{***} (-6.70) 12,841	
F-statistics	174.33	174.15	12,841	12,841 176.47	
Adj. R^2	0.3912	0.3916	0.3919	0.3915	

Notes: This table presents the test of the moderating effect of high-quality auditing on the relation between aggressive tax planning and stock price synchronicity. The independent variables are the four proxies of aggressive tax planning: *ETR**, *LETR**, *BTD*, *DD_BTD*, respectively. The dependent variable is *SYNCH* in year *t*+1. Year and industry fixed effects are included in our models. To alleviate the concern about potential cross-sectional and time-series dependence in the data, we report *t*-values on an adjusted basis, using robust standard errors corrected for double (firm and year) clustering (Gow *et al.*, 2010; Petersen, 2009). *,**,***Significant at the 10, 5 and 1 percent levels, respectively (two-sides)

 Table VII.

 The moderating

 effects of high-quality

 auditing

$$r_{j,t} = \alpha + \beta_{1,j} \times r_{m,t-1} + \beta_{2,j} \times r_{i,t-1} + \beta_{3,j} \times r_{m,t} + \beta_{4,j} \times r_{i,t} + \beta_{5,j} \times r_{m,t+1} + \beta_{6,j} \times r_{i,t+1} + \beta_{7,j} \times r_{m,t-1}^B + \beta_{8,j} \times r_{m,t}^B + \beta_{9,j} \times r_{m,t+1}^B + \varepsilon_{j,t},$$
(12b)

$$r_{j,t} = \alpha + \beta_{1,j} \times r_{m,t-1} + \beta_{2,j} \times r_{i,t-1} + \beta_{3,j} \times r_{m,t} + \beta_{4,j} \times r_{i,t} + \beta_{5,j} \times r_{m,t+1} + \beta_{6,j} \times r_{i,t+1} + \beta_{7,j} \times r_{m,t-1}^{H} + \beta_{8,j} \times r_{m,t}^{H} + \beta_{9,j} \times r_{m,t+1}^{H} + \varepsilon_{j,t},$$
(12c)

where $r_{m,t}^B$ is the value-weighted B-share market return in week *t*, which reflects B-share price movements in both the Shanghai and Shenzhen exchanges. $r_{m,t}^H$ is the value-weighted H-share market return in week *t*, which reflects H-share price movements in both the Shanghai and Shenzhen exchanges. Other variables are as defined earlier. We match the new *SYNCH* (*SYNCH^a*, *SYNCH^b*, *SYNCH^c*) to the firms that issued only A-shares, A+B shares and A+H shares, respectively, the results in Table VIII are consistent with the baseline regression results, in that all the four coefficients are positive and statistically significant.



IJMF 15,5		Column (1)	SYNC Column (2)	Column (4)	
	ETR*	0.0760* (1.83)	_	_	_
	LETR*	- '	0.155*** (2.63)	-	-
	BTD	-	_ ` `	1.011*** (4.92)	-
0.40	DD_BTD	-	-	-	0.460*** (2.72)
848	OTHER CONTRROLS	Yes	Yes	Yes	Yes
	INDUSRY & YEAR	Yes	Yes	Yes	Yes
	INTERCEPT	$-3.618^{***}(-6.44)$	-3.708 *** (-6.56)	$-4.616^{***}(-7.79)$	$-3.529^{***}(-6.31)$
	Observation	13,685	13,685	13,685	13,685
	F-statistics	178.32	178.54	177.84	179.13
	Adj. R^2	0.3627	0.3630	0.3636	0.3629
	Notes: This table prese dependent variable is SI				

only domestic A-shares using Equation (12a), and estimate two different market models for firms with A+B shares and firms with A+H shares using Equations (12b) and (12c). The independent variables are the four proxies of aggressive tax planning: ETR*, LETR*, BTD, DD_BTD, respectively. Year and industry fixed effects are included in our models. We use the OLS model to estimate the regression. To alleviate the concern about potential cross-sectional and time-series dependence in the data, we report t-values on an adjusted basis, using robust standard errors corrected for double (firm and year) clustering (Gow et al., Alternative measures 2010; Petersen, 2009). Variable definitions are in Table AI. *,***Significant at the 10 and 1 percent levels, respectively (two-sides)

5.2 Alternative estimation methods

Table VIII.

of stock price

synchronicity

Table IX presents FFE, Fama and MacBeth (1973) and GMM regression results. Consistent with our baseline results, all the four coefficients on the aggressive tax measures are significantly positive under the FFE specification (Panel A, Table IX). Panel B presents the Fama and MacBeth (1973) regression results that provide standard errors corrected for cross-sectional correlation. The coefficients on all but ETR* tax proxies are positive and significant (e.g. the coefficients on LETR*, BTD and DD_BTD are 0.138 (p < 0.10); 0.99 (p < 0.01) and 0.26 (p < 0.05), respectively). Finally, we re-estimate the parameters of our model using the GMM procedure. We include the lags of the regressors and a constant as instruments, following existing literature (Clarida et al., 1998). The Sargan (Hansen) test indicates that the instrument variables are not weakened or endogenous. For each measure of aggressive tax planning, the *p*-values of the Sargan (Hansen) test of *ETR**, *LETR**, *BTD*, DD BTD are 0.285(0.165), 0.378(0.467), 0.417(0.592), and 0.388(0.135) respectively. The results in Columns (1) to (4) in Panel C of Table IX are qualitatively consistent with our previous results (the coefficient on ETR^* is positive but insignificant, and other measures of aggressive tax planning are positive and significant at 5 percent or better).

5.3 Match sample analysis

Table X reports matched-sample results for the effects of aggressive tax strategies on stock price synchronicity. We employ two different matching methods. First, we match our sample on the basis of firm size, year and industry with no-replacement method. Panel A of Table X presents the size-matched (based on total asset) results. The coefficients on the tax measures are all positive and statistically significant (the coefficients are 0.08, 0.16, 1.01 and 0.56 for ETR*, LETR*, BTD and DD_BTD, respectively, significant at p < 0.05 or better).

We follow the propensity score matching technique developed by Rosenbaum and Rubin (1983). Matching on firm characteristics (covariates) is ideal when the number of characteristics over which the treated and control groups differ is limited. Rosenbaum and Rubin (1983) propose matching by a function of covariates: the probability of an individual



Panel A: FEE DV: SYNCH _{t+1} TAX_AGGR OTHER CONTRROLS INDUSRY& YEAR INSTERCPT Observation F-statistics Within R^2	Column (1) ETR* 0.0834* (1.85) YES YES -6.194*** (-9.00) 13,685 159.10*** 0.3803	Column (2) <i>LETR</i> * 0.235*** (3.26) YES YES -6.352*** (-9.20) 13,685 159.36*** 0.3807	Column (3) <i>BTD</i> 1.078*** (4.56) YES YES -7.413*** (-9.97) 13,685 159.73*** 0.3812	Column (4) DD_BTD 0.441** (2.40) YES YES -6.077*** (-8.86) 13,685 0.3804*** 159.19	Tax planning and stock price synchronicity 849
Panel B: Fama–MacBeth					
DV: $SYNCH_{t+1}$ TAX_AGGR OTHER CONTROLS INTERCEPT Observation Wald χ^2 Adj. R^2	Column (1) <i>LETR</i> * 0.0883 (1.41) Yes -3.663*** (-3.99) 13,685 83.26*** 0.1493	Column (2) <i>ETR</i> * 0.138* (1.92) Yes -3.687*** (-3.98) 13,685 4,844.68*** 0.1490	$\begin{array}{c} \text{Column (3)} \\ BTD \\ 0.989^{**} (2.91) \\ \text{Yes} \\ -4.656^{***} (-4.04) \\ 13,685 \\ 1,519.53^{***} \\ 0.1493 \end{array}$	Column (4) DD_BTD 0.256* (2.30) Yes -3.601*** (-3.83) 13,685 755.44*** 0.1482	
Panel C: GMM					
DV: SYNCH _{t+1} TAX_AGGR OTHER CONTRROLS INDUSRY & YEAR INTERCEPT Observation F-statistics p-value of Arellano-Bond test AR2 p-value of Sargan test p-value of Hansen test	Column (1) ETR^* 0.117 (0.11) Yes -1.589 (-0.49) 10,370 117.59 0.359 0.285 0.165 ()	Column (2) <i>LETR</i> * 0.196* (1.76) Yes -2.186 (-0.66) 10,370 117.20 0.353 0.378 0.467 for large states of the states of	Column (3) BTD 3.594^{***} (2.61) Yes -6.798^{*} (-1.90) 10,370 125.65 0.126 0.417 0.592	Column (4) DD_BTD 0.895** (2.25) Yes -1.925 (-0.62) 10,370 121.80 0.283 0.388 0.135	

Notes: This table presents the robust result of alternative estimation methods. The dependent variable is *SYNCH* in year *t*+1. The independent variables are the four proxies of aggressive tax planning: *ETR**, *LETR**, *BTD*, *DD_BTD*, respectively. Panel A shows the result of firm fixed effects. Panel B shows the result of the Fama–MacBeth regression. Panel C shows the GMM result. As widely used in previous literature, lags of the regressors and a constant are included as instruments (Clarida *et al.*, 1998). *,**,***Significant at the 10, 5 and 1 percent levels, respectively (two-sides)

Table IX. Alternative estimation methods

being selected into the program (treatment group). The following logit regression model was used to estimate firms' propensity to engage in aggressive tax strategies. The dependent variable is a dummy variable coded one for firm-year observations with the respective tax strategies above the sample median, and zero otherwise:

 $Pr(DUMMY_TAX_AGGR_{i,t}) = \alpha_0 + \alpha_2 \times SIZE_{i,t} + \alpha_4 \times STDROA_{i,t} + \alpha_5 \\ \times VOL_{i,t} + \alpha_6 \times LEV_{i,t} + \alpha_7 \times MB_{i,t} + \alpha_8 \\ \times ROA_{i,t} + \alpha_9 \times TOP1_{i,t} + \alpha_{10} \\ \times INSHOLD_{i,t} + \alpha_{11} \times GOVHOLD_{i,t} + \alpha_{12} \\ \times INDNUM_{i,t} + \alpha_{13} \times INDSIZE_{i,t} + \alpha_{14} \\ \times BIG4_{i,t} + \Sigma INDUSTRY + \Sigma YEAR + \varepsilon_{i,t}.$ (13)

We select the optimal match based on the nearest neighbor (NN) technique of the propensity score matching procedure. The NN approach, with replacement, picks a single



IJMF	D	1.1							
15,5	Panel A: closest asset ma	Match by	LTD*	Match by	I ETD*	Match by	DTD	Match by I	מדת תח
10,0	DV: SYNCH _{t+1}	Match by ETK		LET		Match by BTI		DD E	
	TAX_AGGR	0.0796**	-	0.162***		1.009***		0.560*** (3.35)	
	OTHER CONTRROLS	Ves Ves		Yes		Yes		V.500 (5.55) Yes	
	INDUSRY & YEAR	Yes		Yes		Yes		Yes	
	INTERCEPT	-4.149***		-4.081***		-5.034*** (-8.38)		-3.997*** (-7.28)	
850	Observation	13,59		13,576		13,62		13,5	
	<i>F</i> -statistics	191.97		191.21		192.30		192.26	
	Adj. R^2	0.392			0.3926		0.3920		21
	Panel B: Covariate match								
		Match by	ETR*	Match by	LETR*	Match by	r BTD	Match by I	DD_BTD
	Match or not	No	Yes	No	Yes	No	Yes	No	Yes
	Difference	<i>t</i> -test	<i>t</i> -test	<i>t</i> -test	t-test	<i>t</i> -test	<i>t</i> -test	<i>t</i> -test	<i>t</i> -test
	SIZE	5.10***	1.44	3.77***	1.34	4.21***	1.54	5.79***	0.43
	STDROA	6.03***	1.19	3.41***	0.01	5.01***	1.77*	5.27***	2.04**
	VOL	-1.35	-0.38	-3.35***	1.07	1.25***	0.21	1.28	-0.27
	LEV	-5.84^{***}	-1.42	-4.62^{***}	-1.09	-4.10***	-1.29	-3.99***	-1.41
	M/B	6.45***	-1.32	6.00***	-1.22	5.00***	-1.31	4.29***	-1.16
	ROA	4.64***	1.45	5.49***	1.57	3.75***	1.65	4.42***	1.12
	TOP1	0.34	0.09	0.88	-0.09	0.89	-0.11	0.25	-0.04
	INSHOLD	3.18***	-0.47	5.01***	-1.07	3.22***	-0.77	2.04**	-0.15
	GOVHOLD	0.51	0.19	0.91	0.42	1.91	0.56	1.78*	0.98
	INDNUM	1.88*	1.39	1.39	-1.25	1.54	-1.01	1.39	-0.32
	INDSIZE BIG4	1.37 1.53	1.50 -0.82	1.37 3.03***	-1.33 -1.38	1.37 -3.21***	1.10 0.88	1.37 -5.57***	0.13 0.73
	DIG4	1.55	-0.82	3.03	-1.38	-3.21	0.88	-9.97	0.73
	Panel C: PSM regression						DAID		
	DU QUAIQU	Match by		Match by		Match by		Match by I	
	DV: $SYNCH_{t+1}$	ETK		LET.		<i>BT1</i> 1.246***		DD_E	
	TAX_AGGR OTHER CONTRROLS	0.0676 (Yes		0.137** Yes		1.246*** Yes		0.314*	· /
	INDUSRY & YEAR	Yes		Yes		Yes		Ye Ye	
	INDUSKI & ILAK INTERCEPT	-4.450***		-4.248***		-4.914***		-4.064***	
	Observation	-4.450***		-4.246		-4.914		-4.004	
	<i>F</i> -statistics	147.64		146.05		39.06*		11,0	
	Adj. R^2	0.392		0.386		0.381		0.39	
	Notes: This table prese								
Table X. Matching sample analysis	<i>DUM_TAX_AGGR</i> if the and each industry (treat 1 percent levels, respecti	e respective ment group	aggressi), otherv	ive tax meas	ures are	above the sa	ample me	edian across o	each year

control firm according to the closest propensity score. We include a set of firm characteristics that may explain the likelihood that a given firm will be engaged in devising aggressive tax strategies (Cen *et al.*, 2016; Chen *et al.*, 2010). Importantly, the inclusion of these controls ensures a proper balance between treated and untreated subjects in the matched sample, which is one of the key criteria for PSM (Austin, 2011). One important aspect of propensity score matching is to examine the distribution of measured baseline covariates between treated and untreated subjects within the propensity score matched sample. If, after conditioning on the propensity score, no systematic differences exist in baseline covariates between treated and untreated subjects, this indicates that the propensity score model has been correctly specified. In Table X Panel B, none of the included covariates is significantly different between high vs low aggressive tax strategy firms. Panel C, Table X presents the regression results following the PSM approach. The coefficient on *ETR** is positive but insignificant.



However, the coefficients on the remaining tax measures are all positive and significant at	,
p < 0.05 or better.	ar

5.4 Difference-in-difference analysis

In order to investigate whether there is a causal relationship between aggressive tax strategies and price synchronicity, we conduct a difference-in-difference analysis using the 2008 taxation reform in China. In 2008, the Chinese government carried out a taxation reform, which saw a uniform tax rate of 25 percent being imposed on both foreign and domestically funded businesses. For the tax rate increase (decrease) firms, managers are motivated to make aggressive (conservative) tax planning. This policy shock gives us a chance to design a quasi-experiment by designating the tax rate decrease (increase) firms as treatment groups with the "no tax rates" change group as the control group, and then run the following regression:

$$SYNCH_{i,t+1} = \alpha_0 + \alpha_1 \times TAX_INC_{i,t} + \alpha_2 \times TAX_DEC_{i,t} + \alpha_3 \times POST_{i,t} + \alpha_4 \\ \times TAX_INC_{i,t} \times POST_{i,t} + \alpha_5 \times TAX_DEC_{i,t} \times POST_{i,t} + \alpha_i \\ \times CONTROLS_{i,t} + \Sigma INDUSTRY + \Sigma YEAR + \varepsilon_{i,t},$$
(14)

where $TAX_INC_{i,t}$ ($TAX_DEC_{i,t}$) is a dummy variable. For the tax rate increase (decrease) firms, we take it as 1, otherwise 0; $POST_{i,t}$ is the time dummy, taking the value of 1 if the observation comes from 2008, and 0 otherwise. Control variables are the same as those appearing in our baseline regression model. Our variables of interests are α_4 and α_5 . Table XI presents the regression result. The coefficient on the interactive variable $TAX_DEC \times POST$ is negative and statistically significant (coefficient -0.0115, *t*-statistic -2.16, p < 0.05), while that on $TAX_INC \times POST$ is positive and significant (coefficient 0.2144, *t*-statistic 2.03, p < 0.05). The reported results in Table VII, therefore, provide some evidence of a causal relationship between tax strategies and price synchronicity.

6. Conclusion

constant tax rates.

In recent years, a substantial body of literature has investigated the determinants of stock price synchronicity, as well as the economic consequences of aggressive tax planning. But the potential connection between these two factors has not been well examined. Using a large sample of firms listed in China's emerging stock markets, where "investor protection is weak and agency problems are severe" (Allen *et al.*, 2005), we propose and find a positive and

DV	SYNCH _{t+1}
POST	-0.00853** (-2.58)
TAX_DEC	-0.0138** (-2.44)
TAX_INC	0.0226** (1.98)
$POST \times TAX_DEC$	-0.0115** (-2.16)
$POST \times TAX_INC$	0.2144** (2.03)
OTHER CONTROLS	Yes
INDUSTRY	Yes
INTERCEPT	0.00697 (0.21)
Observation	1,285
F-statistics	12.44***
Adj. R^2	0.2803
Notes: This table presents the robustness test: a qua	asi-experiment. Using the sample from 2007 to 2008, we $An \in$
take the tax rate decrease (increase) firms as the tre	atment group, and the control group is the firms with

** *** Significant at the 5 and 1 percent levels, respectively (two-sides)

external shock: tax policy change in China

Table XI.

significant relation between aggressive tax planning and stock price synchronicity. Our results are robust to potential endogeneity and to other specifications, as well as to the use of different tax aggressiveness measures. In addition, we find that the firms engaging in aggressive tax planning exhibit relatively high corporate opacity. Furthermore, we find that the improvement in tax enforcement regime, the stated ownership and the high-quality auditors constrain the adverse effects of tax aggressiveness. These results, taken together, indicate that aggressive firm tax planning reduces the ability of firm stocks to incorporate and reflect information.

Our results add to an understanding of the economic consequences of aggressive tax planning. Although tax strategies help conserve cash by transferring wealth from the government to shareholders, opportunistic managers can extract rent through aggressive tax strategies, and make financial statements less transparent in order to hide such rent extraction, thus, making stock price less informative of firm fundamentals. Our findings, therefore, reinforce the agency perspective of aggressive tax planning, and lend further empirical support to the information-efficiency interpretation of stock price synchronicity. Furthermore, the result of this paper helps us better understand the effects of firm-level tax policy on firm-specific information capitalization in an environment where investor protection (i.e. law/regulatory enforcement and compliance via reporting, auditing and disclosure standards) is relatively weak. Finally, our study has important practical implications for China's regulators, who are striving to reduce the tax burden of enterprises, and also helps investors to better consider investment decisions from a taxation perspective.

Notes

- 1. Other examples of how tax planning strategies can increase opacity, include the creation of entities for multi-state tax planning (e.g. captive real estate investment trusts, intangible holding companies); net operating loss monetization; capital loss utilization, and tax-motivated transfer pricing. Desai (2005) and Desai and Dharmapala (2006) use the Enron and Tyco cases to illustrate how complex structured transactions originating from tax planning can be used to manipulate financial reporting outcomes, because of book-tax nonconformity.
- 2. In the regression model, we replace ETR, LETR by ETR* and LETR*, respectively.
- 3. *INSHOLD* is the sum of the Fund Hold Proportion; QFII Hold Proportion; Broker Hold Proportion; Insurance Hold Proportion; Security Fund Hold Proportion; Entrust Hold Proportion; Finance Hold Proportion and Bank Hold Proportion.
- 4. We use the two-digit code of the CSRC industry classification for the manufacturing industry, and for other industries, we use the one-digit code. In the industry distribution table, the result (61.147 percent) is based on the one-digit code. But in the regression, we control for the manufacturing industry fixed effect based on the two-digit code: C1 (8.13 percent), C2 (19.20 percent), C3 (32.22 percent), and C4 (1.60 percent). We also use the three-digit code for the manufacturing industry, and alternative industry classification standards (WIND: www.wind.com.cn/; SWS: www. swsindex.com/idx0530.asp). The untabulated results are still robust compared with our previous findings. We thank an anonymous referee for this suggestion.

References

- Allen, F., Qian, J. and Qian, M. (2005), "Law, finance, and economic growth in China", Journal of Financial Economics, Vol. 77 No. 1, pp. 57-116.
- Allingham, M.G. and Sandmo, A. (1972), "Income tax evasion: a theoretical analysis", *Journal of Public Economics*, Vol. 1 No. 3, pp. 323-338.
- Armstrong, C.S., Blouin, J.L. and Larcker, D.F. (2012), "The incentives for tax planning", Journal of Accounting and Economics, Vol. 53 Nos 1-2, pp. 391-411.

Austin, P.C. (2011), "An introduction to propensity score methods for reducing the effects of confounding in observational studies", *Multivariate Behavioral Research*, Vol. 46 No. 3, pp. 399-424.



IJMF

15.5

- Balakrishnan, K., Blouin, J. and Guay, W. (2018), "Tax aggressiveness and corporate transparency", *The Accounting Review*, Vol. 94 No. 1, pp. 45-69.
- Baron, R.M. and Kenny, D.A. (1986), "The moderator-mediator variable distinction in social psychological research: conceptual, strategic, and statistical considerations", *Journal of Personality and Social Psychology*, Vol. 51 No. 6, pp. 1173-1182.
- Barrett, M.J. (2004), "'Tax services' as a Trojan horse in the auditor independence provisions of Sarbanes-Oxley", *Michigan State Law Review*, Vol. 463 No. 3, pp. 464-504.
- Beuselinck, C., Deloof, M. and Vanstraelen, A. (2015), "Cross-jurisdictional income shifting and tax enforcement: evidence from public versus private multinationals", *Review of Accounting Studies*, Vol. 20 No. 2, pp. 710-746.
- Box, T. (2018), "Qualitative similarity and stock price co-movement", Journal of Banking & Finance, Vol. 91 No. 6, pp. 49-69.
- Brockman, P. and Yan, X. (2009), "Block ownership and firm-specific information", *Journal of Banking and Finance*, Vol. 33 No. 2, pp. 308-316.
- Cai, H. and Liu, Q. (2009), "Competition and corporate tax avoidance: evidence from Chinese industrial firms", *The Economic Journal*, Vol. 119 No. 537, pp. 764-795.
- Cen, L., Maydew, E.L., Zhang, L. and Zuo, L. (2016), "Customer-supplier relationships and corporate tax avoidance", *Journal of Financial Economics*, Vol. 123 No. 2, pp. 377-394.
- Chan, K. and Hameed, A. (2006), "Stock price synchronicity and analyst coverage in emerging markets", *Journal of Financial Economics*, Vol. 80 No. 1, pp. 115-147.
- Chen, R., Ghoul, S.E., Guedhami, O. and He, W. (2017), "Do state and foreign ownership affect investment efficiency? Evidence from privatizations", *Journal of Corporate Finance*, Vol. 42 No. 2, pp. 408-421.
- Chen, S., Chen, X., Cheng, Q. and Shevlin, T. (2010), "Are family firms more tax aggressive than non-family firms?", *Journal of Financial Economics*, Vol. 95 No. 1, pp. 41-61.
- Cheung, Y.L., Jiang, P., Limpaphayom, P. and Tong, L.U. (2008), "Does corporate governance matter in China?", China Economic Review, Vol. 19 No. 3, pp. 460-479.
- Choudhary, P., Koester, A. and Pawlewicz, R.J. (2017), "Auditor-provided tax services and income tax estimation error", working paper, University of Arizona, Tucson, available at: https://ssrn.com/ abstract=2320479 (accessed April 29, 2019).
- Clarida, R., Gali, J. and Gertler, M. (1998), "Monetary policy rules in practice: some international evidence", *European Economic Review*, Vol. 42 No. 6, pp. 1033-1067.
- Dechow, P.M., Sloan, R.G. and Sweeney, A.P. (1995), "Detecting earnings management", *The Accounting Review*, Vol. 70 No. 2, pp. 193-225.
- Desai, M. and Dharmapala, D. (2006), "Corporate tax avoidance and high-powered incentives", *Journal of Financial Economics*, Vol. 79 No. 1, pp. 145-179.
- Desai, M.A. (2005), "The degradation of reported corporate profits", *Journal of Economic Perspectives*, Vol. 19 No. 4, pp. 171-192.
- Desai, M.A. and Dharmapala, D. (2009), "Corporate tax avoidance and firm value", *Review of Economics & Statistics*, Vol. 91 No. 3, pp. 537-546.
- Desai, M.A., Dyck, A. and Zingales, L. (2007), "Theft and taxes", *Journal of Financial Economics*, Vol. 84 No. 3, pp. 591-623.
- Dhaliwal, D., Gleason, C. and Mills, L. (2004), "Last-chance earnings management: using the tax expense to meet analysts' forecasts", *Contemporary Accounting Research*, Vol. 21 No. 2, pp. 431-459.
- Easley, D. and O'Hara, M. (2004), "Information and the cost of capital", *Journal of Finance*, Vol. 59 No. 4, pp. 1553-1583.
- Fama, E. and MacBeth, J. (1973), "Risk, return, and equilibrium: empirical tests", Journal of Political Economy, Vol. 81 No. 3, pp. 607-636.



853

Tax planning

synchronicity

and stock price

IJMF 15,5	Fan, Z.Y. and Tian, B.B. (2013), "Tax competition, tax enforcement and tax avoidance", <i>Economic Research Journal</i> , Vol. 59 No. 9, pp. 99-111 (in Chinese).
10,0	Fernandes, N. and Ferreira, M.A. (2008), "Insider trading laws and stock price informativeness", <i>Review of Financial Studies</i> , Vol. 22 No. 5, pp. 1845-1887.
	Ferreira, M. and Laux, P. (2007), "Corporate governance, idiosyncratic risk, and information flow", <i>Journal of Finance</i> , Vol. 62 No. 2, pp. 951-989.
854	Frank, M.M. and Rego, S.O. (2006), "Do managers use the valuation allowance account to manage earnings around certain earnings targets?", <i>Journal of the American Taxation Association</i> , Vol. 28 No. 1, pp. 43-65.
	Frank, M.M., Lynch, L.J. and Rego, S.O. (2009), "Tax reporting aggressiveness and its relation to aggressive financial reporting", <i>The Accounting Review</i> , Vol. 84 No. 2, pp. 467-496.
	Gow, I.D., Ormazabal, G. and Taylor, D.J. (2010), "Correcting for cross sectional and time series dependence in accounting research", <i>The Accounting Review</i> , Vol. 85 No. 2, pp. 483-512.
	Gul, F.A., Kim, J.B. and Qiu, A.A. (2010), "Ownership concentration, foreign shareholding, audit quality, and stock price synchronicity: evidence from China", <i>Journal of Financial Economics</i> , Vol. 95 No. 3, pp. 425-442.
	Gupta, S. and Newberry, K. (1997), "Determinants of the variability in corporate effective tax rates: evidence from longitudinal data", <i>Journal of Accounting and Public Policy</i> , Vol. 16 No. 1, pp. 1-34.
	Gupta, S., Laux, R. and Lynch, D. (2016), "Do firms use tax reserves to meet analysts' forecasts? evidence from the pre- and post-FIN 48 periods", <i>Contemporary Accounting Research</i> , Vol. 33 No. 3, pp. 1044-1074.
	Hanlon, M. and Heitzman, S. (2010), "A review of tax research", Journal of Accounting and Economics, Vol. 50 Nos 2-3, pp. 127-178.
	Hanlon, M. and Slemrod, J. (2009), "What does tax aggressiveness signal? Evidence from stock price reactions to news about tax shelter involvement", <i>Journal of Public Economics</i> , Vol. 93 Nos 1-2, pp. 126-141.
	Hanlon, M., Hoopes, J.L. and Shroff, N. (2014), "The effect of tax authority monitoring and enforcement on financial reporting quality", <i>The Journal of the American Taxation Association</i> , Vol. 36 No. 2, pp. 137-170.
	Hao, W., Prevost, A. and Wongchoti, U. (2018), "Are low equity R ² firms more or less transparent? Evidence from the corporate bond market", <i>Financial Management</i> , Vol. 47 No. 4, pp. 865-909.
	Hasan, I., Song, L. and Wachtel, P. (2014), "Institutional development and stock price synchronicity: evidence from China", <i>Journal of Comparative Economics</i> , Vol. 41 No. 1, pp. 92-108.
	Hoopes, J.L., Mescall, D. and Pittman, J.A. (2012), "Do IRS audits deter corporate tax avoidance?", <i>The Accounting Review</i> , Vol. 87 No. 5, pp. 1603-1639.
	Hoopes, J.L., Robinson, L. and Slemrod, J. (2018), "Public tax-return disclosure", <i>Journal of Accounting and Economics</i> , Vol. 66 No. 1, pp. 142-162.
	Hutton, A.P., Marcus, A.J. and Tehranian, H. (2009), "Opaque financial reports, R2, and crash risk", <i>Journal of Financial Economics</i> , Vol. 94 No. 1, pp. 67-86.
	Jiang, X.Y. (2013), "Tax enforcement, tax aggressiveness and stock price crash risk", Nankai Business Review, Vol. 16 No. 5, pp. 152-160 (in Chinese).
	Kanagaretnam, K., Lee, J., Lim, C.Y. and Lobo, G.J. (2016), "Relation between auditor quality and tax aggressiveness: implications of cross-country institutional differences", <i>Auditing: A Journal of</i> <i>Practice and Theory</i> , Vol. 35 No. 4, pp. 105-135.
	Khandaker, S. and Heaney, R. (2008), "Do emerging markets have higher stock synchronicity? The international evidence", <i>Journal of Business and Policy Research</i> , Vol. 8 No. 1, pp. 78-97.
	Khanna, T. and Thomas, C. (2009), "Synchronicity and firm interlocks in an emerging market", Journal of Financial Economics, Vol. 92 No. 2, pp. 182-204.
	Kim, C. and Zhang, L. (2016), "Corporate political connections and tax aggressiveness", <i>Contemporary</i> <i>Accounting Research</i> , Vol. 33 No. 1, pp. 78-114.



- Kim, J.B. and Li, T. (2014), "Multinationals' offshore operations, tax avoidance, and firm-specific information flows: international evidence", Journal of International Financial Management & and stock price Accounting, Vol. 25 No. 1, pp. 38-89.
- Kim, J.-B. and Shi, H. (2012), "IFRS reporting, firm-specific information flows, and institutional environment: international evidence", Review of Accounting Studies, Vol. 17 No. 3, pp. 474-517.
- Kim, J.-B., Li, Y. and Zhang, L. (2011), "Corporate tax avoidance and stock price crash risk; firm-level analysis", Journal of Financial Economics, Vol. 100 No. 3, pp. 639-662.
- Krull, L.K. (2004), "Permanently reinvested foreign earnings, taxes, and earnings management", The Accounting Review, Vol. 79 No. 3, pp. 745-767.
- Kubata, A., Lietz, G.M. and Watrin, C. (2013), "Does corporate tax avoidance impair earnings informativeness?", working paper, University of Muenster, Münster, North Rhine-Westphalia, available at: https://ssrn.com/abstract=2363873 (accessed April 29, 2019).
- Laux, P.A. (2007), "Corporate governance, idiosyncratic risk, and information flow", Journal of Finance, Vol. 62 No. 2, pp. 951-989.
- Li, J. and Myers, S.C. (2006), "R² around the world: new theory and new tests", Journal of Financial Economics, Vol. 79 No. 2, pp. 257-292.
- Li, K., Morck, R., Yang, F. and Yeung, B. (2004), "Firm-specific variation and openness in emerging markets", Review of Economics and Statistics, Vol. 86 No. 3, pp. 658-669.
- Lin, K.J., Karim, K.E. and Carter, C. (2015), "Why does China's stock market have highly synchronous stock price movements? An information supply perspective", Advances in Accounting, Incorporating Advances in International Accounting, Vol. 31 No. 1, pp. 68-79.
- Mertens, J.B. (2003), "Measuring tax effort in central and Eastern Europe", Public Finance & Management, Vol. 3 No. 4, pp. 530-563.
- Mills, L., Erickson, M. and Maydew, E. (1998), "Investments in tax planning", Journal of American Taxation Association, Vol. 20 No. 1, pp. 1-20.
- Minutti-Meza, M. (2013), "Does auditor industry specialization improve audit quality?", Journal of Accounting Research, Vol. 51 No. 4, pp. 779-817.
- Morck, R., Yeung, B. and Yu, W. (2000), "The information content of stock markets: why do emerging markets have synchronous stock price movements?", Journal of Financial Economics, Vol. 58 Nos 1-2, pp. 215-260.
- Nguyen, N.H. and Truong, C. (2013), "The information content of stock markets around the world: a cultural explanation", Journal of International Financial Markets, Institutions and Money, Vol. 26 No. 4, pp. 1-29.
- Pavne, D.M. and Raiborn, C.A. (2018), "Aggressive tax avoidance: a conundrum for stakeholders, governments, and morality", Journal of Business Ethics, Vol. 147 No. 3, pp. 1-19.
- Petersen, M.A. (2009), "Estimating standard errors in finance panel data sets: comparing approaches", Review of Financial Studies, Vol. 22 No. 1, pp. 435-480.
- Phillips, J.D. (2003), "Corporate tax-planning effectiveness: the role of compensation-based incentives", The Accounting Review, Vol. 78 No. 3, pp. 847-874.
- Piotroski, J.D. and Roulstone, D.T. (2004), "The influence of analysts, institutional investors, and insiders on the incorporation of market, industry, and firm-specific information into stock prices", The Accounting Review, Vol. 79 No. 4, pp. 1119-1151.
- Qian, Y. and Roland, G. (1998), "Federalism and the soft budget constraint", American Economic *Review*, Vol. 88 No. 5, pp. 1143-1162.
- Richardson, G., Taylor, G. and Lanis, R. (2013), "The impact of board of director oversight characteristics on corporate tax aggressiveness: an empirical analysis", Journal of Accounting and Public Policy, Vol. 32 No. 3, pp. 68-88.

Roll, R. (1988), "R2", Journal of Finance, Vol. 25 No. 1, pp. 545-566.



855

Tax planning

synchronicity

IJMF 15,5	Rosenbaum, P.R. and Rubin, D.B. (1983), "The central role of the propensity score in observational studies for causal effects", <i>Biometrika</i> , Vol. 70 No. 1, pp. 41-55.
,_	Slemrod, J. (2004), "The economics of corporate tax selfishness", <i>National Tax Journal</i> , Vol. 57 No. 4, pp. 877-899.
	Wilson, R.J. (2009), "An examination of corporate tax shelter participants", <i>The Accounting Review</i> , Vol. 84 No. 3, pp. 969-999.
856	Wu, W., Wu, C., Zhou, C. and Wu, J. (2012), "Political connections, tax benefits and firm performance: evidence from China", <i>Journal of Accounting & Public Policy</i> , Vol. 31 No. 3, pp. 277-300.
	Ye, K.T. and Liu, H. (2011), "Tax enforcement, income tax costs, and earnings management", <i>Management World</i> , Vol. 27 No. 5, pp. 140-148 (in Chinese).
	Zeng, Y.M. and Zhang, J.S. (2009), "Does tax enforcement play a role as corporate governance mechanism", <i>Management World</i> , Vol. 25 No. 3, pp. 143-151 (in Chinese).
	Further reading
	Bramante, R., Petrella, G. and Zappa, D. (2015), "On the use of the market model r-square as a measure of stock price efficiency", <i>Review of Quantitative Finance & Accounting</i> , Vol. 44 No. 2,

- Crocker, K.J. and Slemrod, J. (2005), "Corporate tax evasion with agency costs", Journal of Public Economics, Vol. 89 Nos 9-10, pp. 1593-1610.
- Desai, M.A. (2003), "The divergence between book income and tax income", *Tax Policy & the Economy*, Vol. 17 No. 1, pp. 169-206.
- Durnev, A., Morck, R. and Yeung, B. (2004), "Value-enhancing capital budgeting and firm-specific stock return variation", *The Journal of Finance*, Vol. 59 No. 1, pp. 65-105.
- Goodwin, J. and Wu, D. (2016), "What is the relationship between audit partner busyness and audit quality?", *Contemporary Accounting Research*, Vol. 33 No. 1, pp. 341-377.
- Graham, J.R. (2003), "Taxes and corporate finance: a review", *Review of Financial Studies*, Vol. 16 No. 4, pp. 1075-1129.
- Graham, J.R., Hanlon, M. and Shevlin, T. (2011), "Real effects of accounting rules: evidence from multinational firms' investment location and profit repatriation decisions", *Journal of Accounting Research*, Vol. 49 No. 1, pp. 137-185.
- Gul, F.A. and Yang, Z. (2013), "Do individual auditors affect audit quality? Evidence from archival data", *The Accounting Review*, Vol. 88 No. 6, pp. 1993-2023.
- Khandaker, S. (2011), "R square measure of stock synchronicity", International Review of Business Research Papers, Vol. 7 No. 1, pp. 165-175.
- Khandaker, S. and Heaney, R. (2009), "Do emerging markets have higher stock synchronicity? The international evidence", *Journal of Business and Policy Research*, Vol. 4 No. 1, pp. 79-98.
- Kim, J.-B., Chung, R. and Firth, M. (2003), "Auditor conservatism, asymmetric monitoring, and earnings management", *Contemporary Accounting Research*, Vol. 20 No. 2, pp. 323-359.
- Knechel, W., Vanstraelen, A. and Zerni, M. (2015), "Does the identity of engagement partners matter? An analysis of audit partner reporting decisions", *Contemporary Accounting Research*, Vol. 32 No. 4, pp. 1443-1478.
- Lennox, C.S. and Wu, X. (2017), "A review of the archival literature on audit partners", Accounting Horizons, Vol. 32 No. 2, pp. 1-35.
- Lisowsky, P. (2010), "Seeking shelter: empirically modeling tax shelters using financial statement information", *The Accounting Review*, Vol. 85 No. 5, pp. 1693-1720.
- Morck, R., Yeung, B. and Yu, W. (2013), "R² and the economy", Annual Review of Financial Economics, Vol. 5 No. 1, pp. 143-166.
- Shackelford, D.A. and Shevlin, T. (2001), "Empirical tax research in accounting", Journal of Accounting and Economics, Vol. 31 Nos 1-3, pp. 321-387.



pp. 379-391.

Appendix

Tax planning and stock price synchronicity

he R^2 of the market model in Equation (1) ogarithmic transformation of R^2 for the market model in Equation (1), computed as $\log [R^2/(1-R^2)]$	
ogarithmic transformation of R^2 for the market model in Equation (1), computed as $\log \left[\frac{R^2}{(1-R^2)} \right]$	
	857
Iternative measure of stock price synchronicity. Computed as Equations (12a)–(12c)	007
ffective tax rate	
ong-run cash effective tax rate	
pok-tax difference	
he leverage computed as total liabilities divided by total assets	
arket-to-book ratio, computed as the total market value of equity, divided by the total net assets at	
e end of fiscal year	
he return on asset of the current year, which is defined as earnings before tax divided by total assets	
ternational Big4 audit firms and domestic audit firms and 0 otherwise	
Dechow et al., 1995)	
oxy of corporate opacity. Following Lin et al. (2015), we estimate Real Performance-Total factor	
oductivity (TFP) for each firm year and industry to calculate the degree of private corporate news hiding:	
$(Y) = \alpha + \beta_1 \ln(K) + \beta_2 \ln(L) + \varepsilon,$	
TTESponding industry:	
$IDE_{i,t} = Aos[(Rank_{i,t,k} - Rank_{i,t,k})/(Number of firms in indury n)]$	
no prosty of undator minimutation	
he proxy of individual auditor industry specialization	Table AI.
I V MARY PLANE AND A	Variables definition
hh a chéh ihaéhhhhhhh a nutropero théing 7 hohh	The to-book ratio, computed as the total market value of equity, divided by the total net assets at e end of fiscal year e return on asset of the current year, which is defined as earnings before tax divided by total assets e share proportion of the largest shareholder e percentage of shares held by institutional investors e percentage of shares held by government e percentage of managerial shareholdings e natural log of the number of firms in the industry to which a firm belongs hustry size measured as the log of year-end total assets of all sample firms in the industry to which a n belongs i indicator variable for auditor quality. It equals 1 if a firm is audited by one of the joint ventures of ernational <i>Big4</i> audit firms and domestic audit firms and 0 otherwise exp of corporate opacity. The absolute value of three years accumulated discretionary accruals echow <i>et al.</i> , 1995) oxy of corporate opacity. Following Lin <i>et al.</i> (2015), we estimate Real Performance-Total factor oductivity (TFP) for each firm year and industry to calculate the degree of private corporate news hiding: $(Y) = a + \beta_1 \ln(K) + \beta_2 \ln(L) + e$, ere <i>Y</i> is total income, <i>K</i> is gross property plant and equipment, <i>L</i> is salary. We rank TFP and return on sets (<i>ROA</i>) in the same industry according to the above model and calculate the absolute value of the ference between the two rankings. The greater the absolute value of the difference, the greater the ference between the two rankings. The greater the absolute value of the difference, the greater the ference between the true and the predicted values of a firm's performance, which can reflect the level of vate news hiding. To mitigate the industry effect, we adjusted this value by the number of firms in the responding industry: $DE_{i,i} = Abs[(Rank_{i,k}^{TP} - Rank_{i,k}^{ROA})/(Number of firms in indutry n)]$ e probability of informed trading (PIN) based on the EKO model licator: if the firm is the stated owned companies, it takes value as one, and zero otherwise e level of tax enforce

Corresponding author

Hua Feng can be contacted at: huafeng2-c@my.cityu.edu.hk

For instructions on how to order reprints of this article, please visit our website: www.emeraldgrouppublishing.com/licensing/reprints.htm Or contact us for further details: permissions@emeraldinsight.com



Reproduced with permission of copyright owner. Further reproduction prohibited without permission.

